The convex core of quasifuchsian manifolds with particles

CYRIL LECUIRE JEAN-MARC SCHLENKER

We consider quasifuchsian manifolds with "particles", ie cone singularities of fixed angle less than π going from one connected component of the boundary at infinity to the other. Each connected component of the boundary at infinity is then endowed with a conformal structure marked by the endpoints of the particles. We prove that this defines a homeomorphism between the space of quasifuchsian metrics with nparticles (of fixed angle) and the product of two copies of the Teichmüller space of a surface with n marked points. This extends the Bers double uniformization theorem to quasifuchsian manifolds with "particles".

Quasifuchsian manifolds with particles also have a convex core. Its boundary has a hyperbolic induced metric, with cone singularities at the intersection with the particles, and is pleated along a measured geodesic lamination. We prove that any two hyperbolic metrics with cone singularities (of prescribed angle) can be obtained, and also that any two measured bending laminations, satisfying some obviously necessary conditions, can be obtained, as in Bonahon and Otal [12] in the nonsingular case.

20H10, 57M50

1 Introduction

1.1 Convex cocompact manifolds with particles

Quasifuchsian manifolds A quasifuchsian manifold is a complete hyperbolic manifold M diffeomorphic to $S \times \mathbb{R}$, where S is a closed, oriented surface of genus at least 2, which contains a nonempty, compact, convex subset; see Thurston [54]. Such a manifold has a boundary at infinity, which is the union of two copies of S. Each of those two copies has a conformal structure, τ_+ and τ_- , induced by the hyperbolic metric on M. A celebrated theorem of Bers [1; 8] asserts that the map sending a quasifuchsian metric to (τ_+, τ_-) determines a parametrization of the space of quasifuchsian metrics on M by the product of two copies of the Teichmüller space of S.

A quasifuchsian manifold M contains a smallest nonempty convex subset, called its convex core C(M). Here we say that a subset $K \subset M$ is convex if any geodesic

msp

segment in M with endpoints in K is contained in K. This implies that the inclusion of K in M is a homotopy equivalence.

The boundary of C(M) is again the union of two copies of S, and each is a pleated surface in M, with a hyperbolic induced metric m_+, m_- and a measured bending lamination λ_+, λ_- . (There is a special "Fuchsian" case where C(M) is a totally geodesic surface. The two pleated surfaces mentioned here are then the same, $m_+ = m_-$, and $\lambda_+ = \lambda_- = 0$.)

It is known that any two hyperbolic metrics can be obtained in this way; this follows from Epstein and Marden [26] or from Labourie [37]; however, it is not known whether any couple (m_+, m_-) can be obtained uniquely. Similarly, any two measured laminations λ_+, λ_- can be obtained in this manner if the weight of any leaf is less than π and if λ_+ and λ_- fill *S* (Bonahon and Otal [12]), but it is not known whether uniqueness holds. Recall that λ_- and λ_+ fill *S* if there exists $\epsilon > 0$ such that for any closed curve *c* in *S*, $i(\lambda_-, c) + i(\lambda_+, c) \ge \epsilon$.

The main goal here is to extend those results to quasifuchsian manifolds with "particles", that is, cone singularities of a certain type connecting the two connected components of the boundary at infinity, as described below.

Note that all the results mentioned here are actually known in the more general context of convex cocompact hyperbolic manifolds, ie interiors of compact manifolds with boundary, with a complete hyperbolic metric, containing a nonempty, compact, convex subset (or even more generally for geometrically finite hyperbolic manifolds), the result concerning the measured bending lamination of the boundary can then be found in Lecuire [38]. We stick here to the quasifuchsian setting for simplicity.

Cone-manifolds We consider here hyperbolic cone-manifolds of a special kind, which have cone singularities along curves (a more general notion is defined in [54], allowing for singularities along graphs). Let $\theta \in (0, \pi)$. We call H_{θ}^3 the hyperbolic manifold with cone singularities obtained by isometrically gluing the two faces of a hyperbolic wedge of angle θ (the closed domain in H^3 between two half-planes having the same boundary line). There is a unique such gluing that is the identity on the "axis" of the wedge. We will be using here the following (restrictive) definition.

Definition 1.1 A hyperbolic cone-manifold is a manifold along with a metric for which each point has a neighborhood modeled on H^3_{θ} for some $\theta \in (0, \pi)$.

Let M be a hyperbolic cone-manifold. It has two kind of points: those which have a neighborhood isometric to a neighborhood of a point of some H^3_{θ} outside the cone

singularity are called regular points, and those which are called singular points or cone points. The set of regular points will be denoted by M_r and the set of singular points by M_s . By definition, M_s is a union of curves; if M is complete then those curves can be either closed curves or infinite lines. To each of those curves is associated an angle $\theta \in (0, \pi)$ — such that all points have a neighborhood isometric to a neighborhood of the cone singularity in H_{θ}^3 — which is called its cone angle or simply its angle.

Recall the usual notion of convexity, which differs from other possible notions (eg the local convexity of the boundary of a domain).

Definition 1.2 Let M be a hyperbolic cone-manifold. A subset $K \subset M$ is *convex* if any locally geodesic segment in M with endpoints in K is contained in K.

A nonempty convex subset of M is homotopically equivalent to M and contains all closed geodesics of M; see Moroianu and Schlenker [43, Lemma A.12].

Quasifuchsian manifolds with particles Quasifuchsian manifolds with particles are defined in the same way as nonsingular quasifuchsian manifold.

Definition 1.3 A *quasifuchsian manifold with particles* is a complete hyperbolic conemanifold M isometric to the product $S \times \mathbb{R}$, where S is a closed, orientable surface endowed with a complete hyperbolic metric with cone singularities of angles in $(0, \pi)$ on the lines $\{x_i\} \times \mathbb{R}$ for distinct points x_1, \ldots, x_{n_0} in S, and S contains a nonempty, compact, convex subset. We require that $n_0 \ge 4$ if S is a sphere, ie that M has at least 4 singularities, and that $n_0 \ge 1$ if S is a torus.

Notice that the definition also makes sense if S is a sphere and $n_0 = 3$ but then the metric would be uniquely defined (up to isotopy) by the cone angles. We do not consider this case for technical reasons but also because there is not much to say about it.

Given a nonempty convex subset K of a quasifuchsian manifold with particles, then K contains all closed geodesics of M (see [43, Lemma A.12]) and the inclusion of K in M is a homotopy equivalence (this is proved below).

Geometrically, quasifuchsian manifolds with particles can be considered as I-bundles in the category of hyperbolic 3-manifolds with cone singularities. The term "particle" comes from physical motivations. Quasifuchsian manifolds have Lorentzian siblings called anti-de Sitter (AdS) globally hyperbolic (GH) manifolds, which share many of the key properties recalled above; see Mess [40], and Andersson, Barbot, Benedetti, Bonsante, Goldman, Labourie, Scannell and Schlenker [3]. From a physics viewpoint, GH AdS 3-manifolds are a 3-dimensional toy model for gravity, as they model an empty space with negative cosmological constant. To go beyond an empty model, massive point particles can be added and modeled as cone singularities along timelike lines; see eg 't Hooft [30; 31]. The resulting GH AdS manifolds with particles display some properties that are parallel to those obtained here; see Bonsante and Schlenker [13].

The restrictions on the cone angles — they are supposed to be in $(0, \pi)$ — are necessary at several points here, as they were in [43]. They seem to be physically relevant too. We will mention some points where this hypothesis is useful below as they occur. We do not know whether Theorem 1.7, for instance, can be extended to cone angles less than 2π . In the parallel Lorentzian theory concerning globally hyperbolic anti-de Sitter manifolds, new phenomena arise when the cone angles are larger than π ; see Barbot, Bonsante and Schlenker [5; 6].

Quasifuchsian manifolds with particles are always considered here up to isotopies.

Convex cocompact manifolds with particles The previous definition can be extended to a definition of convex cocompact manifolds with particles.

Definition 1.4 A convex cocompact hyperbolic manifold with particles is a complete hyperbolic cone-manifold M such that:

- M is homeomorphic to the interior of a compact manifold with boundary N.
- The singular locus corresponds under the homeomorphism to a disjoint union of curves in N with endpoints on ∂N .
- The angle at each singular curve is less than π .
- *M* contains a nonempty compact subset that is convex.

A further extension to geometrically finite manifolds with particles is possible; we leave the details to the interested reader. We consider here only quasifuchsian manifolds (with particles) although some of the intermediate statements can be extended to convex cocompact manifolds with particles. There is also some hope to extend the main results to this more general setting, however some technical hurdles have to be overcome before this can be achieved.

Note that there is another possible notion of quasifuchsian manifolds with cone singularities: those which are singular along closed curves, as studied in particular by Bromberg [19; 20]. Although there are similarities between those two kinds conemanifolds (in particular concerning their rigidity), the questions considered here are quite different from those usually associated to those considered for quasifuchsian cone-manifolds with singularities along closed curves (drilling of geodesics etc).

2312

1.2 The conformal structure at infinity

Conformal structures and hyperbolic metrics on surfaces Let us fix some notation.

Definition 1.5 Let *S* be a closed orientable surface, let $x_1, \ldots, x_{n_0} \in S$ be distinct points with $n_0 \ge 4$ if *S* is a sphere and $n_0 \ge 1$ is *S* is a torus, and let $\theta_1, \ldots, \theta_{n_0}$ be in $(0, \pi)$. We then call:

- $\mathcal{T}_{S,x}$ the space of conformal structures on S, considered up to isotopies of S fixing the x_i .
- $\mathcal{H}_{S,x,\theta}$ the space of hyperbolic metrics on S with cone singularities at the x_i , where the angle is θ_i , considered up to isotopies fixing the x_i .

There is a one-to-one map between $\mathcal{T}_{S,x}$ and $\mathcal{H}_{S,x,\theta}$, because any conformal structure contains a unique hyperbolic metric with cone singularities at the x_i of prescribed angle (see Troyanov [56]). We keep distinct notations for clarity.

Notice again that these definitions also make sense when *S* is a sphere and $n_0 = 3$ provided that $\sum_{i=1}^{n_0} \theta_i - 2\pi < 0$. In this case the spaces $\mathcal{T}_{S,x}$ and $\mathcal{H}_{S,x,\theta}$ are points. The statements considered here are already well understood when $n_0 = 0$, so we will focus below on the case $n_0 \ge 1$.

The conformal structure at infinity Non-singular quasifuchsian manifolds have a natural conformal structure at infinity, which can be defined by considering the action of their fundamental group on their discontinuity domain; see [54]. This definition cannot be used directly for quasifuchsian manifolds with particles. However it is still possible to define a conformal structure at infinity; see [43, Section 3.2].

Therefore, to each quasifuchsian metric $g \in Q\mathcal{F}_{\theta_1,...,\theta_{n_0}}$ are associated two points $\tau_+, \tau_- \in \mathcal{T}_{S,n_0}$ corresponding to the conformal structures — marked by the endpoints of the "particles" — on the upper, respectively lower, connected component of the boundary at infinity.

Note that we always implicitly consider conformal structures on the boundary at infinity up to isotopy. (It is therefore not necessary to consider markings.)

A compactness lemma for the conformal structure at infinity We consider again a closed surface *S* along with n_0 distinct points $(n_0 \ge 1) x_1, \ldots, x_{n_0} \in S$ and angles $\theta_1, \ldots, \theta_{n_0} \in (0, \pi)$ so that

$$2\pi\chi(S) - \sum_{i=1}^{n_0} (2\pi - \theta_i) < 0.$$

Geometry & Topology, Volume 18 (2014)

Proposition 1.6 Let $(g_n)_{n \in \mathbb{N}}$ be a sequence of quasifuchsian metrics on $S \times \mathbb{R}$ with particles (cone singularities) on the lines $\{x_i\} \times \mathbb{R}$, of angle equal to θ_i . Suppose that the conformal structures at infinity, $\tau_{-,n}$, $\tau_{+,n} \in \mathcal{T}_{S,n_0}$, converge to conformal structures $\tau_{-,\infty}$, $\tau_{+,\infty}$. Then $(g_n)_{n \in \mathbb{N}}$ has a subsequence converging to a quasifuchsian metric with particles.

The proof is contained in Section 6.2; it is based on the compactness results described below (in Section 3) relative to the induced metric and bending lamination on the boundary of the convex core.

A Bers-type theorem with particles Using the previous proposition, along with the main result of [43], leads to an extension to quasifuchsian manifolds with particles of a classical result of Bers [8] on "double uniformization".

Theorem 1.7 The map from $Q\mathcal{F}_{S,x,\theta}$ to $\mathcal{T}_{S,x} \times \mathcal{T}_{S,x}$ sending a quasifuction hyperbolic metric to the conformal structures at $+\infty$ and at $-\infty$ (marked by the endpoints of the particle) is a homeomorphism.

1.3 The geometry of the convex core

Measured laminations We refer the reader to Casson and Bleiler [22], Penner and Harer [28] and Otal [44] for the definition and main properties of measured laminations on closed (nonsingular) surfaces as well as the topology on the space of measured laminations. There are two possible definitions. One is geometric, in terms of measured geodesic laminations on hyperbolic surfaces with a transverse measure, while the other definition is topological and can involve the boundary at infinity of the universal cover of the surface. The two definitions are equivalent, basically because in a closed (or finite volume) hyperbolic surface any closed curve can be realized uniquely as a closed geodesic. We have the following proposition (see [13, Lemma 2.2] for a proof).

Proposition 1.8 Let Σ be a hyperbolic surface with cone singularities, where the angle is less than π . Let λ be a (topological) lamination on Σ . Then λ can be realized uniquely as a geodesic lamination.

The space of measured geodesic laminations on a hyperbolic surface with cone singularities of angle less than π therefore does not depend on the cone angles.

Definition 1.9 We call \mathcal{ML}_{S,n_0} the space of measured lamination on S with n_0 marked points.

Thus, for any hyperbolic metric m on S with n cone singularities of angle less than π , \mathcal{ML}_{S,n_0} can be canonically identified with the space of measured geodesic laminations on (S,m).

2314

Geometry & Topology, Volume 18 (2014)

Proposition 1.10 Let M be a convex cocompact hyperbolic manifold with particles, and let K and K' be two nonempty convex subsets. Then $K \cap K'$ is a nonempty convex subset.

It leads to a natural definition.

Definition 1.11 Let M be a quasifuchsian manifold with particles. Its *convex core* C(M) is the smallest nonempty convex subset contained in it.

By construction, C(M) is a "minimal" convex subset of M and it follows from general arguments (see [54]) that its boundary is, outside the singular curves, a pleated surface (a locally convex, ruled surface). It turns out that, under the condition that the cone angles are less than π , the boundary of C(M) intersects the cone singularities orthogonally, and is even totally geodesic in the neighborhood of each such intersection; see [43, Lemma A.15].

It follows that there is a well-defined notion of closest-point projection from M to C(M). As a consequence, the inclusion of C(M) in M is a homotopy equivalence. The same holds for any nonempty convex subset of M.

Therefore, given a quasifuchsian metric with particles $g \in Q\mathcal{F}_{S,\theta}$, the induced metrics on the upper and lower boundary components of C(M) (which might coincide in special cases) are two hyperbolic metrics $m_+, m_- \in \mathcal{H}_{S,x,\theta}$. Moreover, those two boundary components are pleated along measured bending laminations $l_+, l_- \in \mathcal{ML}_{S,n_0}$.

A remark on the hypothesis A well-known fact concerning hyperbolic surfaces with cone singularities is that as long as the cone angles are less than π , then any homotopy class of closed curves in the regular part contains a unique geodesic (see, eg, Dryden and Parlier [24]). A fairly direct consequence is that, as for closed surfaces, any topological measured lamination (in the complement of the cone singularities) can be uniquely realized as a measured geodesic lamination.

This is one reason — albeit not the only one — why it is relevant to consider here cone singularities of angle less than π , rather than less than 2π . Indeed for cone singularities of angle less than 2π , the induced metric on the boundary of the convex core might also have cone singularities of angle between π and 2π , and for those metrics the one-to-one relation between measured laminations and measured geodesic laminations is lost.

1.4 Prescribing the bending lamination

Results in the nonsingular case For nonsingular convex cocompact hyperbolic manifolds an existence and uniqueness theorem for metrics with a given rational measured bending lamination was proved by Bonahon and Otal [12]. (Recall that a lamination is rational if its support is a disjoint union of closed curves.) When the lamination is not rational, an existence result was proved in [12] for manifolds with incompressible boundary. It was extended in [38] to manifolds with compressible boundary.

Rational laminations with particles As for quasifuchsian manifolds (without particles), it is possible to give an existence and uniqueness statement concerning the bending lamination on the boundary of the convex core, but only for rational laminations.

Theorem 1.12 Let *S* be a closed orientable surface, let $x_1, \ldots, x_{n_0} \in S$ be distinct points, and let $\theta_1, \ldots, \theta_{n_0}$ be in $(0, \pi)$. Suppose that $n_0 \ge 4$ if *S* is a sphere, and that $n_0 \ge 1$ if *S* is a torus. Let $\lambda_-, \lambda_+ \in \mathcal{ML}_{S,x}$ be measured laminations, each with support a disjoint union of closed curves. Suppose that

- λ_{-} and λ_{+} fill *S*,
- each closed curve in the support of λ_{-} (respectively λ_{+}) has weight less than π .

Then there exists a metric $g \in Q\mathcal{F}_{S,x,\theta}$ such that the measured bending lamination on the upper (respectively lower) boundary component of the convex core of $(S \times \mathbb{R}, g)$ is λ_+ (respectively λ_-). Moreover g is unique up to isotopies.

The proof, which is given in Section 4, is based on the rigidity theorem of Hodgson and Kerckhoff [29] for closed hyperbolic manifolds with cone singularities. We prove in Lemma 4.3 that the hypothesis are necessary conditions.

General laminations When considering laminations which are not necessarily rational, we obtain only a weaker result, because we can only claim existence, but not uniqueness (this remains an open problem even in the nonsingular case; see [12; 38]).

Theorem 1.13 Let *S* be a closed surface, let $x_1, \ldots, x_{n_0} \in S$ be distinct points, and let $\theta_1, \ldots, \theta_{n_0}$ be in $(0, \pi)$. Let $\lambda_-, \lambda_+ \in \mathcal{ML}_{S,x}$. Suppose that

- λ_{-} and λ_{+} fill *S*,
- each closed curve in the support of λ_- (respectively λ_+) has weight less than π .

Then there exists a metric $g \in Q\mathcal{F}_{S,x,\theta}$ such that the measured pleating lamination on the upper (respectively lower) boundary component of the convex core of $(S \times \mathbb{R}, g)$ is λ_+ (respectively λ_-).

The two conditions on λ_- , λ_+ in this theorem are easily seen to be necessary when g is not Fuchsian; see Lemma 4.3. Note that both Theorem 1.12 and Theorem 1.13 are restricted to quasifuchsian manifolds with particles, rather than more general convex cocompact manifolds with particles.

1.5 The induced metric on the boundary of the convex core

The Bers-type result on the conformal metric at infinity can be used to obtain an existence result concerning the prescription of the induced metric on the boundary of the convex core.

Theorem 1.14 Let $m_{-}, m_{+} \in \mathcal{H}_{S,n_{0},\theta}$, where $\theta = (\theta_{1}, \ldots, \theta_{n_{0}}) \in (0, \pi)^{n}$. There exists a quasifuchsian metric with particles on $S \times \mathbb{R}$, with particles of angle θ_{i} at the lines $\{x_{i}\} \times \mathbb{R}$, for which the induced metric on the boundary of the two connected components of the convex core are m_{-} and m_{+} .

In the smooth case — ie for quasifuchsian hyperbolic manifolds without conical singularities — the corresponding result is well known; it follows either from results of Labourie [37] or from a partial answer, first given by Epstein and Marden [26], to a conjecture of Sullivan. (The conjecture made by Sullivan turned out to be wrong — see Epstein and Markovic [27] — but the result proved by Epstein and Marden is sufficient to prove Theorem 1.14 in the nonsingular context.)

As for the conformal structure at infinity, it might be possible to extend this statement to cover convex cocompact (respectively geometrically finite) manifolds with particles. The uniqueness remains elusive, as in the nonsingular case.

1.6 Applications

Quasifuchsian manifolds can be used as tools in Teichmüller theory. By extension, the quasifuchsian manifolds with particles considered here can be used as tools for the study of the Teichmüller space of hyperbolic metrics with cone singularities (of angle less than π) on a surface.

One such application is through the renormalized volume of those quasifuchsian manifolds with particles, as considered in Krasnov and Schlenker [33; 35]. In the nonsingular case this renormalized volume is equal to the Liouville functional (see Takhtajan and Zograf [52; 53] and Takhtajan and Teo [51]); it is a Kähler potential on $\mathcal{H}_{S,x,\theta}$. Other applications of closely related tools, in the nonsingular context, for the global geometry of the Weil–Petersson metric on Teichmüller space, can be found in McMullen [39]. Yet other applications to some properties of the grafting map are considered in Krasnov and Schlenker [34], and the manifolds with particles considered here should allow for an extension to the grafting map on $\mathcal{H}_{S,x,\theta}$.

1.7 Outline of the proofs

We now turn to a description of the main technical points of the proofs.

Measured bending laminations Theorem 1.12 is proved by an argument strongly influenced by the proof given by Bonahon and Otal [12] for nonsingular convex cocompact manifolds. Thanks to a doubling trick, the infinitesimal rigidity of the convex cores of convex cocompact manifolds with particles, with respect to the (rational) measured bending lamination, is reduced to an important infinitesimal rigidity result proved for hyperbolic cone-manifolds by Hodgson and Kerckhoff [29]. A deformation argument then provides the proof of the theorem.

The existence result for general laminations on quasifuchsian manifolds with particles (Theorem 1.13) can then be obtained by an approximation argument, as in the nonsingular case in [12; 38]. The key step of the proof is a compactness statement, showing that if the measured bending laminations converge to a limit having good properties, then the quasifuchsian metrics converge after extracting a subsequence. However the arguments developed in [12; 38] cannot be used in the context of quasifuchsian manifolds with particles, because they rely heavily on the representation of the fundamental group. Different arguments are therefore used here, which are more differential-geometric in nature.

Those arguments are sometimes technically involved because of the added difficulties induced by the particles. However, after stripping the proof of the elements which are needed only because of the particles (for instance the multiple cover argument used in Section 3.4 to find simplicial surfaces with given boundary in the convex core), the compactness proof given here is simpler than the one in [12; 38].

Prescribing the induced metric on the boundary of the convex core We give in Section 6 a rather elementary proof of Theorem 1.14, which has two parts. Call t_- (respectively t_+) the hyperbolic metric in the conformal class τ_- (respectively τ_+) with cone angles θ_i at x_i . The first part is an upper bound on the length of the curves in the hyperbolic metric at infinity t_{\pm} , following Bridgeman [15], and Bridgeman and Canary [16].

Now recall that, by Thurston's earthquake theorem (Kerckhoff [32] and Thurston [55]), there exists a unique right earthquake sending a given hyperbolic metric to another one. This extends to hyperbolic metrics with cone singularities of angle less than π ; see [13]. In particular there is a unique measured lamination ν_+ such that the right earthquake along ν_+ , applied to m_+ , yields the hyperbolic metric t_+ . The second part of our proof is a bound on the length of ν_+ for m_+ (see Proposition 5.2).

2318

This is then used in Section 6 to prove Theorems 1.7 and 1.14. The proof of Theorem 1.7 also uses another main ingredient, the local rigidity of quasifuchsian manifolds with particles proved in [43].

Quasi-conformal estimates There is another possible way to prove Theorem 1.14, closer to the argument used in the nonsingular case (as seen in [15; 16; 26]). It uses a bound on the quasi-conformal factor between the conformal structure at infinity τ and the conformal class of the induced metric *m* on the boundary of the convex core, both understood as elements of \mathcal{T}_{S,n_0} .

Proposition 1.15 There exists a constant C > 0 (depending only on the topology of *M*) such that τ is *C*-quasiconformal to *m*.

This proposition is not formally necessary to obtain the main results presented here; its proof can be found in the appendix.

As mentioned above, the proof of Theorem 1.14 through Proposition 1.15 would be much closer to the proof(s) known in the nonsingular case. It can be pointed out that the proof given in Section 6 is quite parallel, but in the context of Teichmüller theory understood as the study of hyperbolic rather than complex surfaces. From this viewpoint, Proposition 5.2 is a direct analog of Proposition 1.15, with quasiconformal deformations replaced by earthquakes.

What follows Section 2 presents the definition of the convex core of a convex cocompact manifold with particles, and some of its simple properties, extending well-known properties with no cone singularity. In Section 3 we state and prove a key compactness statement with respect to the measured bending lamination on the boundary of the convex core. Section 4 contains the proof of Theorem 1.12, using a local rigidity statement of Hodgson and Kerckhoff [29] and the compactness lemma of Section 3. Section 5 contains the proof of Theorem 1.13, and Section 6 contains the proof of Theorem 1.7 and of Theorem 1.14. Section 7 contains some remarks on the analogy with corresponding problems in anti-de Sitter geometry and on applications to the Weil–Petersson metric of the Teichmüller space of hyperbolic metrics with cone singularities of prescribed angles on a closed surface (see [33; 35; 34]). Finally, the appendix contains the proofs of Proposition 1.15, based on the estimates on the length of the earthquake lamination obtained in Section 5.

2 The geometry of the convex core

This section contains some basic statements necessary to understand the geometry of convex cocompact manifolds with particles, concerning in particular the convex core

and its boundary. We consider here such a convex cocompact manifold with particles, M, and denote by M_r its regular part and by M_s its singular part (the union of the singular lines).

We exclude below the simplest case where M is Fuchsian, that is, where it is the warped product of hyperbolic surfaces with cone singularities (S, h) by \mathbb{R} , with the metric $dt^2 + \cosh(t)^2 h$. In this Fuchsian case the convex core is a surface, corresponding to t = 0, and it is totally geodesic outside the intersection with the particles, and orthogonal to those particles.

2.1 Surfaces orthogonal to the singular locus

We define here a natural notion of pleated surface orthogonal to the singular locus in M. The first step is to define the notion of totally geodesic plane orthogonal to a cone singularity in a hyperbolic cone-manifold. The first condition is that the surface is totally geodesic outside its intersections with the particles. The second condition is local, in the neighborhood of the intersections with the particles; there, the surface should correspond to the image in H^3_{θ} of the restriction to the wedge (used to define H^3_{θ}) of a plane orthogonal to the axis of the wedge.

Definition 2.1 Let Σ be a pleated surface in M_r and let Σ' be its closure as a subset of M; suppose that $\Sigma' \setminus \Sigma \subset M_s$. We say that Σ' is *orthogonal to the singular locus* if any $x \in \Sigma' \setminus \Sigma$ has a neighborhood in Σ' which is a totally geodesic surface orthogonal to the singular locus.

This definition can be extended to encompass more general surfaces, ie surfaces which are neither pleated nor totally geodesic in the neighborhood of the singular locus. In this more general case the definition can be given in terms of the convergence of the unit normal vector to a vector "tangent" to the singular locus at its intersection with the surface. This will however not be needed here.

2.2 The convex core of a manifold with particles

Among the defining properties of a quasifuchsian cone-manifold M is the fact that it contains a compact subset K which is convex in the (strong) sense that any geodesic segment in M with endpoints in K is contained in K. We have already seen that it is possible to define the *convex core* of M as the smallest compact subset of M which is convex, denoted by C(M).

Theorem 2.2 Suppose that C(M) is not a totally geodesic surface. Then its boundary is the disjoint union of surfaces which are orthogonal to the singular locus. Each connected component of the singular locus of M intersects C(M) along a segment.

The proof is a consequence of two lemmas, both stated under the hypothesis of the theorem. The second lemma in particular gives more precise information on the geometry of the convex core; it is taken from [43, Lemma A.14].

Lemma 2.3 The boundary of C(M) is a surface orthogonal to the singular locus.

Let $x \in M$. We denote by L_x the *link* of M at x, that is, the space of directions of geodesic rays starting from x, with its natural angle distance. When x is a regular point of M, L_x is isometric to the 2-dimensional sphere S^2 with its round metric. When x is contained in a singular line of angle θ , L_x can be described as the metric completion of the quotient by a rotation of angle θ of the universal cover of the complement of two antipodal points in S^2 .

Definition 2.4 Let $K \subset M$ be convex, and let $x \in M$. The *link* of K at x is the set of directions $v \in L_x$ such that there is a (small) geodesic ray starting from x in the direction v which is contained in K. It is denoted by $L_x(K)$.

Clearly $L_x(K) = \emptyset$ when x is not contained in K, while $L_x(K) = L_x$ when x is contained in the interior of K.

To go further, we define the oriented normal bundle of $\partial C(M)$, denoted by $N_r^1 \partial C(M)$, as the set of $(x, n) \in TM$ such that $x \in \partial C(M)$ is not in the singular locus of M and n is a unit vector such that its orthogonal is a support plane of C(M) at x, and n is oriented towards the exterior of C(M).

Let $x \in M$ be a nonsingular point, let $v \in T_x M$ and let $t \in \mathbb{R}_+$. For t small enough, it is possible to define the image of (x, tv) by the exponential map: it is the point $\exp(x, tv) := g(t)$, where g is the geodesic, parametrized at constant speed, such that g(0)x and g'(0) = v. As t grows, $\exp(x, tv)$ remains well-defined until g intersects the singular set of M.

Lemma 2.5 The exponential map is a homeomorphism from $N_r^1 \partial C(M) \times (0, \infty)$ to the complement of C(M) in M_r . The map

$$\exp_{\infty} \colon N_r^1 \partial C(M) \to \partial_{\infty} M,$$
$$(x, v) \mapsto \lim_{t \to \infty} \exp(x, tv)$$

is a homeomorphism from $N_r^1 \partial C(M)$ to the complement in $\partial_{\infty} M$ of the endpoints of the singular curves in M.

Geometry & Topology, Volume 18 (2014)

This follows directly from [43, Lemma A.11].

The proof of Theorem 2.2 clearly follows from Lemma 2.3 and Lemma 2.5, since Lemma 2.5 shows that the cone singularities cannot re-enter the convex core after exiting it.

2.3 The geometry of the boundary

By construction, C(M) is a minimal convex set in M, and it follows as in the nonsingular case (see [54]) that its boundary is a "pleated surface" except at its intersections with the singular curves.

Lemma 2.6 The surface $\partial C(M)$ has an induced metric that is hyperbolic (ie it has constant curvature -1) with conical singularities at the intersections of $\partial C(M)$ with the singular curves of M, where the total angle is the same as the total angle around the corresponding singular curve. It is "pleated" along a measured lamination λ in the complement of the singular points. Moreover the distance between the support of λ and the intersection of the singular set of M with $\partial C(M)$ is strictly positive.

Proof Since C(M) is a minimal convex subset, its boundary is locally convex and ruled, therefore developable (see Spivak [50, Chapter 3] for the Euclidean analogue or Thurston [54]) so that its induced metric is hyperbolic. The fact that its intersection points are conical singularities, with a total angle which is the same as the total angle around the corresponding singularities, is a consequence of the fact that $\partial C(M)$ is orthogonal to the singularities.

Similarly, the fact that $\partial C(M)$ is pleated along a measured lamination is a direct consequence of the fact that it is ruled and locally convex, ie that each point in $\partial C(M)$ is in either a complete hyperbolic geodesics or a totally geodesic ideal triangle. The support of λ is a disjoint union of embedded maximal geodesics, and it is well known (see [24]) that (under the hypothesis that the angles at the cone singularities are strictly less than π) embedded geodesics remain at positive distance from the singular locus. So the distance between the support of λ and the singular locus of $\partial C(M)$ is strictly positive.

2.4 The distance between the singular curves

We state and prove here some elementary statements on the distance between singular points in the boundary of C(M) and between singular curves in M. They will be useful at several points below.

2322

Lemma 2.7 Let $\theta \in (0, \pi)$. There exist $\epsilon > 0$ and $\rho > 0$, depending on θ , such that:

- (1) In a complete hyperbolic surface with cone singularities of angles less than θ (not homeomorphic to a sphere with three singularities), two cone singularities are at distance at least ϵ from each other.
- (2) If D is a closed 2-dimensional geodesic disk of radius ε centered at a singular point x₀ of cone angle θ, and if Ω ⊂ D is a convex subset whose closure intersects the boundary of D, then Ω contains all points of D at distance at most ρ from x₀.

In particular, it follows from point (1) that no embedded geodesic in D can come within distance less than ρ from the cone singularity.

Proof The first point is well known; see [24]. The interested reader can construct an elementary proof based on Dirichlet domains, as in 3–dimensional manifolds in the proof of the second point below.

For the second point, let $x_1 \in \partial D \cap \Omega$ and let γ be the minimizing geodesic segment from x_0 to x_1 . Since D contains no other singular point by the first point, the complement of γ in D is isometric to an angular sector in the disk of radius ϵ in H^2 . This angular sector has three vertices; one corresponding to x_0 and the other two corresponding to x_1 . Since $\theta < \pi$, it is convex at the vertex corresponding to x_0 . Let s be the geodesic segment joining the two vertices corresponding to x_1 . Then Ω , being convex, contains the projection in D of the triangle bounded by s and by the two geodesic segments in the boundary of D joining x_0 to the two vertices projecting to x_1 . This proves the statement, with ρ equal to the distance between x_0 and s. \Box

We now turn to a similar lemma, but concerning 3–dimensional manifolds with particles. Although we include a short proof, it is a consequence of Boileau, Leeb and Porti [10, Theorem 5.3].

Lemma 2.8 Let $\theta, \theta' \in (0, \pi)$. There exists $\epsilon > 0$, depending on θ , such that, if M is a quasifuchsian manifold with particles of angles more than θ' and less than θ , then any two particles in M are at distance at least ϵ from each other.

Proof We reason by contradiction, that is, we fix $\theta \in (0, \pi)$ and assume that for any n > 0 there is a quasifuchsian manifold M_n with particles of angles less than θ , and a length minimizing segment s_n with length less than 1/n joining two particles p_n and p'_n or one particle p_n with itself and no length minimizing segment shorter than s_n . Let x_n be the midpoint of s_n . We call D_n the Dirichlet domain in M_n centered at x_n .

We call (M'_n, x_n) the pointed cone-manifold obtained by performing on (M_n, x_n) a homothety of ratio $1/L(s_n)$, so that M'_n has constant curvature $L(s_n)^2 \le 1/n^2$. Let D'_n be the Dirichlet domain centered at x_n in M'_n , so that D'_n is obtained by performing a homothety of ratio $1/L(s_n)$ on D_n .

By construction the cone singularities in M'_n are at distance at least 1, so that, after extracting a subsequence, (M'_n, x_n) converges in the pointed Gromov-Hausdorff topology to a pointed manifold (M', x). Still by construction, M' is a noncompact orientable Euclidean cone 3-manifold and either M' contains at least one cone singularity pwith cone angle less than π , the limit of p_n , and a geodesic loop limit of s_n or M'contains at least two cone singularities p and p' with cone angle less than π , the limits, respectively, of p_n and p'_n . Let D' be the Dirichlet domain centered at x in M'. Then (D', x) is the limit of the (D'_n, x_n) in the pointed Gromov-Hausdorff topology. By definition all D'_n are unbounded, so D' is also unbounded.

It follows from Thurston's classification of noncompact orientable Euclidean cone 3-manifolds (see Boileau and Porti [11] or Cooper, Hodgson and Kerckhoff [23]) that the only possibility for M' is a thick turnover $S^2(\alpha, \beta, \gamma) \times \mathbb{R}$, where $S^2(\alpha, \beta, \gamma)$ is a 2-sphere with 3 singular points with angles (α, β, γ) such that $\alpha + \beta + \gamma = 2\pi$. Therefore, for *n* large enough M_n is homeomorphic to the product of a sphere by a line with three singularities. But this is impossible since the M_n are quasifuchsian manifolds with particles, and the definition explicitly excludes such manifolds.

We now call $\epsilon_0 > 0$ the number ϵ associated by the previous two lemmas to the maximum of the θ_i , and ρ_0 the corresponding value of ρ .

3 Compactness statements

3.1 Main statement

The main goal of this section is to prove the following compactness lemma.

Lemma 3.1 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Let λ_n be the measured bending laminations on the boundary of the convex core of M_n . Suppose that $\lambda_n \rightarrow \lambda_\infty$, where λ_∞ satisfies the hypothesis of Theorem 1.13. Then, after taking a subsequence, M_n converges to a quasifuchsian manifold with particles of the common topological type, the limit particles and measured bending lamination λ_∞ . Let us explain the definitions used in this statement. The *topological type* of a quasifuchsian manifold with particles M has the form $(S, x_1, \ldots, x_{n_0})$, where S is a compact surface with genus at least 2 and x_1, \ldots, x_{n_0} are distinct points on S. A quasifuchsian manifold with particles M has topological type $(S, x_1, \ldots, x_{n_0})$ if M is isometric to the product $S \times \mathbb{R}$ endowed with a complete hyperbolic metric with cone singularities of angles $\theta^i \in (0, \pi)$ on the lines $\{x_i\} \times \mathbb{R}$.

Consider a sequence of quasifuchsian manifolds M_n with particles with the same topological type $(S, x_1, \ldots, x_{n_0})$. Denote by $\theta_n^i \in (0, \pi)$ the cone angles of the metric of M_n on $\{x_i\} \times \mathbb{R}$. Then the sequence M_n has *converging angles* if and only if θ_n^i converges in $(0, \pi)$ when n goes to ∞ for any $i \leq n_0$.

Notice that since the λ_n converge to λ_{∞} , λ_n is eventually nontrivial. In particular the manifolds M_n are not Fuchsian (ie their convex cores are not surfaces) except maybe for finitely many of them. Throughout this section, when we consider a quasifuchsian manifold with particles, we will assume that it is not Fuchsian so that its convex core is a 3-dimensional manifold with boundary.

The convex core C(M) of M is homeomorphic to $S \times I$. Thus $\partial C(M)$ is homeomorphic to $S \sqcup S$ and each copy of S in this union has n_0 marked points x_1, \ldots, x_{n_0} corresponding to the endpoints of the particles. The measured bending lamination on the boundary of C(M) is an element of the space

$$\mathcal{ML}_{S,n_0} \times \mathcal{ML}_{S,n_0}$$

of measured laminations on two copies of S with n_0 marked points. The space \mathcal{ML}_{S,n_0} is endowed with the topology of weak-* convergence of measures on compact transversals and $\mathcal{ML}_{S,n_0} \times \mathcal{ML}_{S,n_0}$ is endowed with the product topology. In simple terms, we can fix a finite but sufficiently large set of curves c_i that are either closed curves or segments between two singular points; then, two measured laminations are close if and only if their intersection with each of the c_i are close.

Although Lemma 3.1 is a generalization of the "lemme de fermeture" of [12], the proof is very different. The reason is that the two main ingredients of the proof in Bonahon and Otal's paper are the Culler–Morgan–Shalen compactification of the character variety by actions on \mathbb{R} –trees and the covering theorem of Canary. Since both these results hardly extend to manifolds with particles we had to use different arguments. Since our proof also works without particles, we get a new proof of the main result of [12].

3.2 A finite cover argument

We work under the assumption that the cone angle around each singularity is less than π . This assumption guarantees that the singularities are never too close to each

other (see Lemma 2.8) and that the boundary of the convex core is well-defined and is orthogonal to the singularities. On the other hand, cone singularities with cone angles less than π can be viewed as singularities with concentrated positive curvature. But some of the results we will use are easier to prove when the curvature is negative. To overcome this difficulty, we will use a branched cover for which the cone angles are all greater than 2π .

Let M be a quasifuchsian manifold with particles. A branched cover $\overline{M} \to M$ branched along the singularities is *negatively curved* if the cone angles around the singularities of the metric induced on \overline{M} are all greater than 2π . We call \overline{M} a *negatively curved branched cover of* M.

This name comes from the fact that a singularity with cone angle greater than 2π can be viewed as a set of concentrated negative curvature. More precisely \overline{M} can be approximated by Riemannian manifolds with curvature bounded above by -1 (in the bilipschitz topology). It follows that \overline{M} has properties of negatively curved manifolds, in particular the uniqueness of the geodesic segment joining two given points in a given homotopy class.

We will construct such branched covers for sequences. Consider a sequence of quasifuchsian manifold with particles M_n with the same topological type $(S, x_1, \ldots, x_{n_0})$ (as defined in the preceding section). We denote by g_n the metric of M_n and by θ_n^i the cone angles of g_n on the component κ_i of the singular locus corresponding to x_i . Assume that the sequence M_n has converging angles, namely θ_n^i converge to some $\theta^i \in (0, \pi)$ for any $i \le n_0$.

For each singularity x_i , we choose an integer k_i such that $2\pi/k_i$ is less than the angle θ^i (the limit of θ^i_n). The surface S with cone angle $2\pi/k_i$ at the point x_i is a hyperbolic orbifold. As such it has a manifold cover $h: \overline{S} \to S$ that is a branched cover so that the lifts of the point x_i have a branching index equal to k_i . The branched cover $h: \overline{S} \to S$ extends naturally to a branched cover $h: \overline{S} \times I \to S \times I$.

For a fixed *n*, we have the metric g_n on $S \times \mathbb{R}$ with cone singularities θ_n^i along $\{x_i\} \times \mathbb{R}$. If we pull back g_n using the map *h*, we get a hyperbolic metric $\overline{g_n}$ with cone singularities on $\overline{S} \times \mathbb{R}$ for which the covering transformations are isometries. Let $\overline{M}_n = (\overline{S} \times \mathbb{R}, \overline{g_n})$ be the manifold with cone singularities thus obtained. By the choice of $\{k_1, \ldots, k_{n_0}\}$, for *n* large enough, we have $k_i \theta_n^i \ge 2\pi$, hence the cone angle of $\overline{g_n}$ around each singularity of \overline{M} is at least 2π . Thus for *n* large enough, \overline{M}_n is a negatively curved branched cover of M_n and the topological type of \overline{M}_n does not depend on *n*.

2326

3.3 Pleated annuli

A technical device that will be useful later on is a simplicial annulus bounded by two given curves. As was mentioned above, when we consider a quasifuchsian manifold with particles, we assume that it is not Fuchsian.

Let us first fix some notations. We consider a quasifuchsian manifold with particles M with topological type $(S, x_1, \ldots, x_{n_0})$. We denote by g the complete hyperbolic metric with cone singularities of M and by C(M) the convex core of M. We will use a negatively curved branched cover \overline{M} of M (as defined in the previous section). The construction of such a cover is explained above for a sequence M_n , here we take the constant sequence, $M_n = M$ for any n, to define \overline{M} . We denote by $C(\overline{M}) \subset \overline{M}$ the preimage of the convex core C(M) of M under the covering projection, by $\overline{\lambda}$ its bending measured geodesic lamination and by \overline{m} the induced metric on $\partial C(\overline{M})$. We will use this notation throughout this section.

Now let us construct our simplicial annulus.

Lemma 3.2 Let M be a quasifuchsian manifold with particles and \overline{M} a negatively curved branched cover of M. Let $\overline{d}, \overline{d'}$ be homotopic simple closed geodesics, respectively on the upper and on the lower boundary components of $C(\overline{M})$. There exists an immersed annulus \overline{A} in $C(\overline{M})$ bounded by $\overline{d} \cup \overline{d'} \subset \partial C(\overline{M})$ such that the metric induced on \overline{A} by g_n is a hyperbolic metric with cone singularities with angles at least 2π . The area of \overline{A} is at most $\min\{l_{\overline{m}}(\overline{d}) + l_{\overline{m}}(\overline{d'}), i(\overline{\lambda}, \overline{d}) + i(\overline{\lambda}, \overline{d'})\}$.

Proof Let us specify that $l_{\overline{m}}(\overline{d})$, respectively $l_{\overline{m}}(\overline{d}')$, is the length of \overline{d} , respectively \overline{d}' , with respect to the metric \overline{m} induced by \overline{g} on $\partial C(\overline{M})$.

Since \overline{d} and $\overline{d'}$ are disjoint homotopic simple closed curves, there is an embedded annulus $\overline{A} \subset C(\overline{M})$ with $\partial \overline{A} = \overline{d} \cup \overline{d'}$. If the bending laminination of $C(\overline{M})$ intersects \overline{d} and $\overline{d'}$ finitely many times, then \overline{d} and $\overline{d'}$ are piecewise geodesics. If not, we approximate them by piecewise geodesic curves and work on the approximations. Consider a triangulation T of \overline{A} whose vertices are all contained in $\overline{d} \cup \overline{d'}$ and such that any vertex of \overline{d} and $\overline{d'}$ (when considered as piecewise geodesics) is a vertex of T. As we have said before, in \overline{M} , there is a unique geodesic segment joining 2 given points in a given homotopy class. It follows that we can change \overline{A} by a homotopy so that each edge of T is a geodesic segment in $C(\overline{M})$. Next, for each triangle T_i of T, we choose a vertex v and we substitute T_i by the geodesic cone from v to the edge e_v of T_i not containing v. This geodesic cone is the union of the geodesic segments joining v to the edge e_v of T_i (the homotopy class of such segment is defined by the corresponding segment of T_i). Again the existence of this cone follows from the uniqueness of geodesic paths. From now on we denote this cone by T_i . By construction, it is a locally ruled surface and as such has negative curvature:

Claim 3.3 Let *M* be a hyperbolic manifold with cone singularities with cone angles bigger than 2π . Given a point $v \in M$ and a geodesic segment $e_v \subset M$, a geodesic cone $T_i \subset M$ from v to e_v is an union of polygons with curvature -1. Furthermore, the sum of the angles of the polygons meeting at an interior vertex is at least 2π .

Proof The surface T_i meets the singular locus \overline{M}_s of \overline{M} along segments and at points. For each component κ of $\overline{M}_s \cap T_i$ we consider the two extremal segments joining v to e_v and intersecting κ . Doing this for each component of $\overline{M}_s \cap T_i$, we get a family of segments which are geodesic for the metric of $(\overline{M}, \overline{g})$ and hence for the induced metric on T_i . We add the components of $\overline{M}_s \cap T_i$ that are segments to this family and get a new family of geodesic segments. The closure of each complementary region is a polygon, ie a disc with piecewise geodesic boundary (see Figure 1). By construction each such polygon is a locally ruled surface in \mathbb{H}^3 hence it has curvature -1. Thus we have proved the first sentence of this claim.



Figure 1: Decomposition of T_i into hyperbolic polygons

By construction, given an interior vertex v of this decomposition into polygons, there is a geodesic segment (for the metric of $(\overline{M}, \overline{g})$) which passes through v. On each side of this segment, the sum of the angles of the polygons has to be at least π . Thus, we can conclude that the sum of the angles of the polygons around v is at least 2π . \Box

We change the annulus \overline{A} so that it is a union of geodesic cones as described in Claim 3.3. Thus the induced metric is hyperbolic with cone singularities with angles greater than 2π . By the Gauss–Bonnet formula, the area of \overline{A} is at most the bending of $\partial \overline{A}$, namely Area $(\overline{A}) \leq i(\partial \overline{A}, \overline{\lambda}_n) = i(\overline{d}, \overline{\lambda}_n) + i(\overline{d}', \overline{\lambda}_n)$.

It remains to prove that $\operatorname{Area}(\overline{A}) \leq l_{\overline{m}}(\overline{d}) + l_{\overline{m}}(\overline{d}')$. By construction, \overline{A} is a union of triangles T_i such that one edge of each T_i lies in $\overline{d} \cup \overline{d}'$ and by Claim 3.3, the induced metric on each such triangle is a hyperbolic metric with cone singularities with angles greater than 2π . It follows that the induced metric can be approximated by Riemannian metrics with curvature at most -1. Let T_i^h be a hyperbolic triangle (ie a geodesic triangle in \mathbb{H}^2) such that the length of the edges of T_i^h are the same as the length of the edges of T_i . Since the induced metric on T_i as curvature at most -1, we have $\operatorname{Area}(T_i) \leq \operatorname{Area}(T_i^h)$. On the other hand the area of a hyperbolic triangle is less than the length of any of its edges (see [54, Lemma 9.3.2]). It follows that $\operatorname{Area}(T_i)$ is less than the length of any of its edges; in particular it is less than the length of the edge of T_i lying in $\overline{d} \cup \overline{d'}$. Since this holds for all the triangles composing \overline{A} , we have $\operatorname{Area}(\overline{A}) \leq l_{\overline{m}}(\overline{d}) + l_{\overline{m}}(\overline{d'})$.

3.4 Long geodesics in M

In this section, we will show that, under the hypothesis of Lemma 3.1, the induced metrics on $\partial C(M_n)$ are bounded. In order to do that we will show that if some geodesic is long in the boundary of C(M), then the boundary of some annulus is almost not bent or the bending lamination tends to have a leaf with a weight greater than or equal to π . Since this would contradict the conditions on λ_{∞} , it will follow that any given simple closed curve on $\partial C(M_n)$ has bounded length. As earlier, when we consider a quasifuchsian manifold with particles, we assume that it is not Fuchsian.

Throughout this section we use the following notation. We have a sequence of quasifuchsian manifolds with particles M_n with the same topological type $(S, x_1, \ldots, x_{n_0})$. We denote by $(g_n)_{n \in \mathbb{N}}$ the metric of M_n . We assume that the sequence M_n has converging angles, namely the θ_n^i converge in $(0, \pi)$ for any $i \leq n_0$. Since M_n is quasifuchsian (with particles), $C(M_n)$ is homeomorphic to $S \times I$. We denote by S and S' the two components of $\partial C(M_n)$. S' is homeomorphic to S. Let m_n be the metric defined on $S \sqcup S'$ by the identification with $\partial C(M_n)$ endowed with the metric induce by the g_n -length of paths. This metric m_n is a hyperbolic metric with cone singularities of angles θ_n^i at the points $\{x_i\} \in S$ and $\{x'_i\} \in S'$. We denote by $\lambda_n \in \mathcal{ML}_{S,n_0} \times \mathcal{ML}_{S,n_0}$ the bending measured geodesic lamination of $\partial C(M_n)$.

We will make use of the branched covers \overline{M}_n defined in Section 3.2. Recall that, for *n* large enough, \overline{M}_n is a negatively curved branched cover of M_n and that the topological type of \overline{M}_n does not depend on *n*. We denote by $C(\overline{M}_n)$, \overline{m}_n and $\overline{\lambda}_n$ the preimages of $C(M_n)$, m_n and λ_n under the covering map $\overline{M}_n \to M_n$.

We will use the next lemma to prove that, under the right hypothesis on λ_{∞} , the induced metric on $\partial C(M_n)$ is bounded.

Lemma 3.4 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Assume that (λ_n) converges to λ_{∞} (without any hypothesis on λ_{∞}). Consider a simple closed curve $d \subset S$. Let $d_n \subset \partial C(M_n)$ be the closed m_n -geodesic freely homotopic to d. If $l_{m_n}(d_n) \to \infty$, then either λ_{∞} contains a leaf with a weight greater than or equal to π , or there is a sequence of essential annuli E_n such that $i(\lambda_n, \partial E_n) \to 0$.

Proof Let S' be the other boundary component of C(M) (ie not S) and let d'_n be the closed m_n -geodesic freely homotopic to d lying in S'. Let \overline{M}_n be a negatively curved branched cover of M_n so that for n large enough the topological type of \overline{M}_n does not depend on n. Let \overline{d}_n and $\overline{d}'_n \subset \partial C(\overline{M}_n)$ be homotopic lifts of d_n and d'_n respectively under the covering projection $\overline{M} \to M$. The preimage $\overline{\lambda}_n \in \mathcal{ML}(\partial \overline{M})$ of λ_n is the bending measured lamination of $C(\overline{M}_n)$. Furthermore $\overline{\lambda}_n$ converges to the preimage $\overline{\lambda}_\infty$ of λ_∞ .

First we will show that if \overline{d}_n is long compared to the area of an annulus \overline{A}_n bounded by $\overline{d}_n \cup \overline{d}'_n$ then there are shortcuts in \overline{A}_n . Namely $\overline{d}_n \cup \overline{d}'_n$ contains points that are close to each other in \overline{M}_n but far in $\overline{d}_n \cup \overline{d}'_n$. This can happen for instance if d_n and d'_n are close to each other in $C(M_n)$.

Claim 3.5 Let M_n be a sequence of quasifuchsian manifolds with particles and let \overline{M}_n be a negatively curved branched cover of M_n such that the topological type of \overline{M}_n does not depend on n. Assume that $(\overline{\lambda}_n)_{n \in \mathbb{N}}$ converges to $\overline{\lambda}_\infty$. Let $\overline{d} \subset \overline{S}$ be a simple closed curve and denote by $\overline{d}_n \subset \overline{S} \subset \partial C(\overline{M}_n)$ the simple closed \overline{m}_n -geodesic in the homotopy class of \overline{d} .

If $l_{\overline{m}_n}(d_n) \to \infty$, then there is a geodesic arc $\overline{k}_n \subset C(\overline{M}_n)$ such that $l_{g_n}(\overline{k}_n) \to 0$ and that either \overline{k}_n joins the two components of $\partial C(\overline{M}_n)$ or the \overline{m}_n -geodesic arc $\overline{\kappa}_n \subset \overline{d}_n$ in the homotopy class of \overline{k}_n relative to its boundary satisfies $l_{\overline{m}_n}(\overline{\kappa}_n) \to \infty$.

Proof Denote by \overline{d}'_n the closed \overline{m}_n -geodesic lying in $\overline{S}' \subset \partial C(\overline{M}_n)$ that is homotopic to \overline{d}_n in $C(\overline{M}_n)$. Consider the annulus \overline{A}_n with $\partial \overline{A}_n = \overline{d}_n \cup \overline{d}'_n$ that was constructed in Lemma 3.2. Since $l_{\overline{m}_n}(\overline{d}_n) \to \infty$, there is $\varepsilon_n \to 0$ and a segment $\overline{s}_n \subset \overline{d}_n$ such that $l_{\overline{m}_n}(\overline{s}_n) \to \infty$ and $i(\overline{s}_n, \overline{\lambda}_n) \leq \varepsilon_n$. Let $\overline{t}_n \subset C(\overline{M}_n)$ be the \overline{g}_n -geodesic segment homotopic to \overline{s}_n relative to its endpoints. Since \overline{s}_n is almost not bent, its length is very close to the length of \overline{t}_n (see [38, Lemme A2]). In particular, $l_{\overline{g}_n}(\overline{t}_n) \to \infty$. Furthermore, for the same reason, any point in \overline{t}_n is close to \overline{s}_n . Namely there is $\eta_n = \eta(\varepsilon_n) \to 0$ such that for any point $\overline{z}_n \subset \overline{t}_n$, there is $\overline{x}_n \subset \overline{s}_n$ with $d_{\overline{g}_n}(\overline{x}_n, \overline{z}_n) \leq \eta_n$ (see [38, Affirmation A3]). Since $(\overline{\lambda}_n)_{n \in \mathbb{N}}$ converges to $\overline{\lambda}_{\infty}$, the bending $i(\overline{\lambda}_n, \partial \overline{A}_n)$ of $\partial \overline{A}_n$ converges. By Lemma 3.2 the area of \overline{A}_n is bounded. Now, in \overline{d}_n , we replace \overline{s}_n by \overline{t}_n . If needed, we approximate \overline{d}'_n and this new curve \overline{d}_n by piecewise geodesic curves and consider an annulus \overline{A}_n between \overline{d}_n and \overline{d}'_n with a triangulation that respects the piecewise geodesic structures of $\overline{d_n}$ and $\overline{d'_n}$ (as described in Lemma 3.2). We change $\overline{A_n}$ so that it is a union of geodesic cones. It follows from the assumption that $(\lambda_n)_{n \in \mathbb{N}}$ converges to λ_{∞} , from Lemma 3.2 and from the previous paragraph that Area $(\overline{A_n})$ is a bounded sequence. For any point in $\overline{t_n}$ that is at distance at least $\frac{1}{3}l_{\overline{g_n}}(\overline{t_n})$ from $\partial \overline{t_n}$, we consider in \overline{A}_n an arc orthogonal to \overline{t}_n that either hits $\partial \overline{A}_n$ at distance less than η_n from its basepoint or has length η_n (η_n will be specified later on). Let $\overline{Z}_n \subset \overline{A}_n$ be the union of those arcs that have length η_n and let \overline{z}_n be the union of their starting points (ie their intersection with $\overline{t_n}$). The set $\overline{Z_n}$ is embedded and its area is the same as the area of a strip of length $l_{\overline{g}_n}(\overline{z}_n)$ and width η_n . Notice that since the singularities of A_n have cone angles at least 2π , the area of this strip is at least the area of a hyperbolic strip with the same length and width, it is at least $l_{\overline{g}_n}(\overline{z}_n) \sinh(\eta_n)$. Let K be a number larger than the area of \overline{A}_n for any *n*. Taking η_n such that $\sinh(\eta_n) > 3K/l_{\overline{g}_n}(\overline{t}_n)$, we get

$$K \ge \operatorname{Area}(\overline{Z}_n) \ge l_{\overline{g}_n}(\overline{z}_n) \sinh(\eta_n) > K \frac{3l_{\overline{g}_n}(\overline{z}_n)}{l_{\overline{g}_n}(\overline{t}_n)}.$$

Hence $l_{\overline{g}_n}(\overline{z}_n) < \frac{1}{3}l_{\overline{g}_n}(\overline{t}_n)$. It follows that there exists an arc with length less than η_n orthogonal to \overline{t}_n whose starting point $\overline{x}'_n \subset \overline{t}_n$ is at a distance at least $\frac{1}{3}l_{\overline{g}_n}(\overline{t}_n)$ from $\partial \overline{t}_n$ (distance measured on \overline{t}_n) and which hits $\partial \overline{A}_n$ in a point $\overline{y}_n \subset \partial \overline{A}_n$.

As we have seen in the previous paragraph there is a point $\overline{x}_n \in \overline{s}_n$ very close to \overline{x}'_n . It follows that $\overline{x}_n \in \overline{d}_n$ and $\overline{y}_n \in \overline{d}_n \cup \overline{d}'_n$ are joined in $C(\overline{M}_n)$ by an arc \overline{k}_n satisfying $l_{g_n}(\overline{k}_n) \to 0$.

If $\overline{y}_n \in \overline{d}'_n$ then we are done. Otherwise \overline{x}_n and \overline{y}_n both lie in \overline{d}_n . By construction \overline{k}_n lies in an annulus connecting \overline{d}_n to \overline{d}'_n . It follows that there is a \overline{m}_n -geodesic arc $\overline{\kappa}_n \subset \overline{d}_n$ that is homotopic to \overline{k}_n relative to $\{\overline{x}_n\} \cup \{\overline{y}_n\}$. Since \overline{x}_n is at distance at least $l_{\overline{m}_n}(\overline{t}_n)/3$ from the points in $\partial \overline{s}_n$, we have

$$l_{\overline{m}_n}(\overline{\kappa}_n) \ge \frac{l_{\overline{m}_n}(\overline{t}_n)}{3} \to \infty.$$

Consider the points \overline{x}_n and \overline{y}_n constructed in Claim 3.5 and extract a subsequence such that either $\overline{y}_n \in \overline{d}_n$ for any n or $\overline{y}_n \in \overline{d}'_n$ for any n. We will show below that if \overline{y}_n lies in \overline{d}_n then λ_∞ has a leaf with a weight greater than or equal to π and that if \overline{y}_n lies in \overline{d}'_n then there is a sequence of essential annuli $E_n \subset M$ such that $i(\lambda_n, \partial E_n) \to 0$. In the next step we are going to construct \overline{m}_n -geodesic loops based at \overline{x}_n and \overline{y}_n that are almost not bent.

Claim 3.6 Let M_n be a sequence of quasifuchsian manifolds with particles and let \overline{M}_n be a negatively curved branched cover of M_n such that the topological type of \overline{M}_n does not depend on n. Consider 2 points \overline{x}_n , $\overline{y}_n \in \partial C(\overline{M}_n)$ away from the singularities and a \overline{g}_n -geodesic arc $\overline{k}_n \subset C(\overline{M}_n)$ joining \overline{x}_n to \overline{y}_n such that $l_{\overline{g}_n}(\overline{k}_n) \to 0$ and that either \overline{x}_n and \overline{y}_n lie on different components of $\partial C(M_n)$ or there is a \overline{m}_n -geodesic arc $\overline{k}_n \in \partial C(\overline{M}_n)$ that is homotopic to \overline{k}_n relative to its boundary $\{\overline{x}_n\} \cup \{\overline{y}_n\}$ and that satisfies $l_{\overline{m}_n}(\overline{k}_n) \to \infty$.

Consider a loop \overline{l}_n on $\partial C(\overline{M}_n)$ based at \overline{x}_n , which is geodesic for \overline{m}_n (except at \overline{x}_n). Let $\overline{f}_n \subset \partial C(\overline{M}_n)$ be the \overline{m}_n -geodesic loop based at \overline{y}_n that is homotopic to \overline{l}_n . Assume that $l_{\overline{m}_n}(\overline{l}_n)$ is bounded. Then the bending of \overline{l}_n and \overline{f}_n tends to 0, namely $i(\overline{l}_n, \overline{\lambda}_n) \to 0$ and $i(\overline{f}_n, \overline{\lambda}_n) \to 0$.

Proof When saying that \overline{x}_n and \overline{y}_n are away from the singularities we mean that there is a uniform lower bound on their distance to the singular locus of \overline{M}_n .

Let \widetilde{M}_n be the universal cover of \overline{M}_n : it is a simply connected hyperbolic 3-manifold with cone singularities. Let $C(\widetilde{M}_n)$ be the lift of $C(\overline{M}_n)$ to \widetilde{M}_n . Let $\tilde{l}_n, \tilde{k}_n, \tilde{x}_n, \tilde{y}_n$ be lifts of $\overline{l}_n, \overline{k}_n, \overline{x}_n$ and \overline{y}_n with $\widetilde{x}_n \in \tilde{l}_n$ and $\widetilde{x}_n \cup \widetilde{y}_n = \partial \widetilde{k}_n$. The point $\overline{x}'_n = \partial \widetilde{l}_n \setminus \widetilde{x}_n$ is the image of \overline{x}_n under a covering transformation. Consider the \widetilde{m}_n -geodesic arc $\widetilde{f}_n \subset \partial C(\widetilde{M}_n)$ joining \overline{y}_n to its image \overline{y}'_n under this covering transformation.

Let us first assume that there are no singularities in \overline{M}_n . Then \widetilde{M}_n is isometric to \mathbb{H}^3 and we choose the isometry so that \widetilde{x}_n is identified with a fixed point of \mathbb{H}^3 (independently of *n*). Let $\Pi(\widetilde{x}_n)$ be a support plane for $C(\widetilde{M}_n)$ at \widetilde{x}_n , namely a totally geodesic plane that intersects $C(\widetilde{M}_n)$ only along $\partial C(\widetilde{M}_n)$ and contains \widetilde{x}_n . Up to moving \overline{x}_n slightly, we may assume that it is disjoint from $\overline{\lambda}_n$ so that there is only one support plane at \widetilde{x}_n . The convex set $C(\widetilde{M}_n)$ lies in a half-space $E(\widetilde{x}_n)$ bounded by $\Pi(\widetilde{x}_n)$. Similarly let $\Pi(\widetilde{y}_n)$ be a support plane at \widetilde{y}_n and let $E(\widetilde{y}_n)$ be the half-space bounded by $\Pi(\widetilde{y}_n)$ that contains $C(\widetilde{M}_n)$.

If \overline{x}_n and \overline{y}_n lie on different components of $\partial C(M_n)$ then $\Pi(\widetilde{x}_n)$ and $\Pi(\widetilde{y}_n)$ are disjoint.

If \overline{x}_n and \overline{y}_n lie on the same component of $\partial C(M_n)$, since $l_{\overline{m}_n}(\overline{\kappa}_n) \to \infty$, either $\Pi(\widetilde{x}_n)$ and $\Pi(\widetilde{y}_n)$ are disjoint or their intersection goes to ∞ with *n* (namely the sequence $\Pi(\widetilde{x}_n) \cap \Pi(\widetilde{y}_n)$ lies outside larger and larger compact sets in \mathbb{H}^3).

In both cases, since $l_{\overline{g}_n}(\overline{k}_n) \to 0$, \widetilde{y}_n converges to \widetilde{x}_n (viewed as a fixed point in \mathbb{H}^3) and, up to extracting a subsequence, $\Pi(\widetilde{x}_n)$ and $\Pi(\widetilde{y}_n)$ converges to the same plane Π_{∞} in \mathbb{H}^3 . Furthermore $E(\widetilde{x}_n)$ converges to a half-space $E(\widetilde{x}_{\infty})$ bounded by Π_{∞} and $E(\widetilde{y}_n)$ converges to the other half-space $E(\widetilde{y}_{\infty})$ bounded by Π_{∞} . Let $\Pi(\tilde{x}'_n)$ be a support plane at \tilde{x}'_n . Since $l_{\bar{m}_n}(\bar{f}_n)$ is bounded, up to extracting a subsequence, \tilde{x}'_n converges in \mathbb{H}^3 . Again since either \bar{x}'_n and \bar{y}_n lie on different components of $\partial C(M_n)$ or $d_{\tilde{m}_n}(\tilde{x}'_n, \tilde{y}_n) \to \infty$, $\Pi(\tilde{y}_n) \cap \Pi(\tilde{x}'_n)$ either is empty or goes to infinity. It follows that $\Pi(\tilde{x}'_n)$ also converges to Π_∞ and that $E(\tilde{x}'_n)$ converges to $E(\tilde{x}_\infty)$. The external dihedral angle between $\Pi(\tilde{x}_n)$ and $\Pi(\tilde{x}'_n)$ is an upper bound for $i(\bar{l}_n, \bar{\lambda}_n)$, hence $i(\bar{l}_n, \bar{\lambda}_n) \to 0$.

It remains to show that $i(\overline{f_n}, \overline{\lambda}_n) \to 0$.

By construction, $d_{\tilde{g}_n}(\tilde{x}'_n, \tilde{y}'_n) = d_{\tilde{g}_n}(\tilde{x}'_n, \tilde{y}'_n) = \ell_{\bar{g}_n}(\bar{k}_n) \to 0$, hence $\Pi(\tilde{x}'_n)$ and $\Pi(\tilde{y}'_n)$ converge to the same plane Π_{∞} in \mathbb{H}^3 . If \bar{x}_n and \bar{y}_n lie on different components of $\partial C(M_n)$ then $\Pi(\tilde{x}'_n)$ and $\Pi(\tilde{y}'_n)$ are disjoint. It follows that $E(\tilde{y}'_n)$ converges to $E(\tilde{y}_{\infty})$ while $E(\tilde{x}'_n)$ converges to $E(\tilde{x}_{\infty})$, which implies $i(\bar{f}_n, \bar{\lambda}_n) \to 0$ as above. If \bar{x}_n and \bar{y}_n lie on the same component of $\partial C(M_n)$ then

$$d_{\widetilde{m}_n}(\widetilde{x}'_n, \widetilde{y}'_n) = d_{\widetilde{m}_n}(\widetilde{x}_n, \widetilde{y}_n) \to \infty.$$

It follows that $E(\tilde{y}'_n)$ converge to $E(\tilde{y}_{\infty})$, which again implies $i(\bar{f}_n, \bar{\lambda}_n) \to 0$.

When the \overline{M}_n have singularities, we cannot define support planes, but we can define local support planes at points that are disjoint from the singularities. Thus we can locally use the same arguments as in the nonsingular case, leading to the same conclusion. \Box

Let us choose for \overline{l}_n a shortest \overline{m}_n -geodesic loop based at \overline{x}_n . Since the area of $(\overline{S} \sqcup \overline{S}', \overline{m}_n)$ is bounded, there is a constant Q > 0 such that $\ell_{\overline{m}_n}(\overline{l}_n) \leq Q$. By Claim 3.6, we have $i(\overline{l}_n, \overline{\lambda}_n) \to 0$. Let \overline{f}_n be the \overline{m}_n -geodesic loop based at \overline{y}_n that is homotopic to \overline{l}_n . By Claim 3.6, we have $i(\overline{f}_n, \overline{\lambda}_n) \to 0$. Since \overline{l}_n and \overline{f}_n are freely homotopic in $C(\overline{M}_n)$, there is an annulus \overline{E}_n bounded by \overline{l}_n and \overline{f}_n .

If \overline{y}_n lies in \overline{d}'_n , then \overline{l}_n and \overline{f}_n lie in different components of $\partial \overline{M}$. In particular \overline{E}_n is an essential annulus for any n. Furthermore, we have $i(\overline{\lambda}_n, \partial \overline{E}_n) \leq i(\overline{l}_n, \overline{\lambda}_n) + i(\overline{f}_n, \overline{\lambda}_n)$, and the second member tends to 0.

Consider the projection E_n of \overline{E}_n to $C(M_n)$. Although E_n may not be embedded, it follows from the annulus theorem [57] that any neighborhood of E_n contains an embedded annulus, which we still denote by E_n . We have then $i(\lambda_n, \partial E_n) \to 0$.

Thus we have proved:

Claim 3.7 Let M_n be a converging sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Consider a geodesic arc $k_n \subset C(M_n)$ joining the two components of $\partial C(M_n)$ such that $\ell_{m_n}(k_n) \to 0$. Then there is a sequence of essential annuli E_n such that, up to extracting a subsequence, $i(\lambda_n, \partial E_n) \to 0$.

If \overline{y}_n lies in \overline{d}_n then \overline{l}_n and \overline{f}_n are homotopic on $\partial C(\overline{M}_n)$. We are going to show that in this case λ_n tends to have a leaf with a weight greater than or equal to π .

Claim 3.8 Let M_n be a sequence of quasifuchsian manifolds with converging angles and let \overline{M}_n be a negatively curved branched cover of M_n such that the topological type of \overline{M}_n does not depend on n. Consider a \overline{g}_n -geodesic arc $\overline{k}_n \subset C(\overline{M}_n)$ such that $\ell_{\overline{g}_n}(\overline{k}_n) \to 0$ and that there is a \overline{m}_n -geodesic arc $\overline{\kappa}_n \in \partial C(\overline{M}_n)$ that is homotopic to \overline{k}_n relative to its boundary and that satisfies $\ell_{\overline{m}_n}(\overline{\kappa}_n) \to \infty$. Then $\liminf i(\overline{\kappa}_n, \overline{\lambda}_n) \ge \pi$.

Proof The curve $\overline{k}_n \cup \overline{k}_n$ is a skew polygon (up to approximating \overline{k}_n by piecewise geodesic segments) and bounds a disc in $C(\overline{M}_n)$. Consider the geodesic cone \overline{D}_n from \overline{x}_n to $\overline{k}_n \cup \overline{k}_n$. As in the proof of Lemma 3.2, the induced metric on \overline{D}_n is a hyperbolic metric with cone singularities with cone angles of at least 2π . Since \overline{k}_n is short, the local support planes at the endpoints \overline{x}_n and \overline{y}_n of \overline{k}_n are close to each other (compare with the proof of Claim 3.6). It follows that the sum of the internal angles of \overline{D}_n at \overline{x}_n and \overline{y}_n is close to being greater than π , namely there is $\varepsilon_n \to 0$ such that the sum of these 2 angles is greater than $\pi - \varepsilon_n$. Now the Gauss–Bonnet formula shows that lim inf $i(\overline{k}_n, \overline{\lambda}_n) \ge \pi$.

Using this claim, we will now show that under the right hypothesis, λ_n tends to have a leaf with a weight greater than or equal to π .

Claim 3.9 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type $S \times I$ and converging angles. Let $d \subset S$ be a simple closed curve and consider its geodesic representative d_n on one component of $\partial C(M_n)$. Consider an arc $\kappa_n \subset d_n$ and denote by k_n the geodesic arc in M_n in the homotopy class of κ_n relative to its boundary. If $\ell_{g_n}(k_n) \to 0$, $\ell_{m_n}(\kappa_n) \to \infty$ and λ_n converges, its limit λ_∞ has a leaf with a weight greater than or equal to π .

Proof Let \overline{M}_n be a negatively curved branched cover of M_n whose topological type does not depend on n. Let $\overline{\kappa}_n$ and \overline{k}_n be lifts of κ_n and k_n respectively. Let \overline{x}_n and \overline{y}_n be the endpoints of $\overline{\kappa}_n$ and let $\overline{l}_n \subset \partial C(\overline{M}_n)$ be a shortest geodesic loop based at \overline{x}_n . Let $\overline{f}_n \subset \partial C(M_n)$ be the geodesic loop based at \overline{y}_n that is homotopic to \overline{l}_n on $\partial C(\overline{M}_n)$. Let \widetilde{S}_n be the universal cover of the connected component of $\partial C(\overline{M}_n)$ containing \overline{x}_n , endowed with the induced metric. Pick a connected component $\overline{l}_n \subset \widetilde{S}_n$ of the preimage of \overline{l}_n under the covering projection. This broken geodesic \widetilde{l}_n is invariant under a primitive covering transformation γ_n , and we denote by \widetilde{f}_n the component of the preimage of \overline{f}_n that is also invariant under γ_n . The lines \widetilde{l}_n and \widetilde{f}_n are disjoint and bound an infinite band \widetilde{B}_n ; they are connected by a lift $\widetilde{\kappa}_n$ of $\overline{\kappa}_n$, and by its translates by $\gamma_n^k, k \in \mathbb{Z}$. Pick a simple closed geodesic $\overline{e}_n \subset \partial C(\overline{M}_n)$ and let \widetilde{e}_n be a lift of \overline{e}_n to \widetilde{S}_n . It is easy to check that a component \widetilde{a}_n of $\widetilde{e}_n \cap \widetilde{B}_n$, which is an arc connecting \widetilde{l}_n to \widetilde{f}_n , satisfies $i(\widetilde{a}_n, \widetilde{\lambda}_n) \ge i(\widetilde{\lambda}_n, \widetilde{\kappa}_n) - \#\{\widetilde{a}_n \cap \bigcup_k g_n^k \widetilde{\kappa}_n\}(i(\overline{l}_n, \overline{\lambda}_n) + i(\overline{f}_n, \overline{\lambda}_n))$, where $\widetilde{\lambda}_n$ is the preimage of $\overline{\lambda}_n$ under the covering projection. By Claim 3.6, $i(\overline{l}_n, \overline{\lambda}_n) \to 0$ and $i(\overline{f}_n, \overline{\lambda}_n) \to 0$, and by Claim 3.8, $\liminf i(\overline{\kappa}_n, \overline{\lambda}_n) \ge \pi$. Notice that $\#\{\widetilde{a}_n \cap \bigcup_k g_n^k \widetilde{\kappa}_n\}$ is bounded except if \overline{e}_n spirals more and more toward \overline{c}_n . If, for example, we assume that \overline{e}_n and \overline{c}_n converge to intersecting geodesic laminations, \overline{e}_n does not spiral toward \overline{c}_n and we find that $\liminf i(\overline{e}_n, \overline{\lambda}_n) \ge i(\overline{e}_n, \overline{c}_n)\pi$.

We will now use this inequality to conclude that, when the $\overline{\lambda}_n$ converge, its limit $\overline{\lambda}_{\infty}$ has a leaf with a weight greater than or equal to π .

Let us notice that, up to extracting a subsequence, the homotopy class of \bar{c}_n does not depend on n. Otherwise there is a simple closed curve $\bar{e} \subset \bar{S}$ such that $i(\bar{e}, \bar{c}_n) \to \infty$. To see that, extract a subsequence such that the \bar{c}_n converge in the Hausdorff topology, pick a simple closed curve \bar{e} that intersects this limit transversally and apply the inequality above to $\{\bar{e}_n\} = \{\bar{e}\}$. But $i(\bar{e}, \bar{c}_n) \to \infty$ would contradict the assumption that $\bar{\lambda}_n$ converges.

Let $\overline{c} \subset S$ be a simple closed curve in the homotopy class defined by \overline{c}_n . By the inequality above, we have $\liminf i(\overline{e}, \overline{\lambda}_n) \ge i(\overline{e}, \overline{c})\pi$ for any simple closed curve \overline{e} . It follows easily that \overline{c} is a leaf of $\overline{\lambda}_n$ with a weight greater than or equal to π . Taking the quotient, we conclude that λ_{∞} has a leaf with a weight greater than or equal to π . \Box

It is now easy to conclude the proof of Lemma 3.4. Under the assumptions of Lemma 3.4, namely when there is a simple closed curve d such that $\ell_{m_n}(d_n) \to \infty$, it follows from Claims 3.5, 3.7 and 3.9 that either there is a sequence of essential annuli $E_n \subset C(M_n)$ such that $i(\lambda_n, \partial A_n) \to 0$ or λ_∞ contains a leaf with a weight equal to at least π . \Box

We can now deduce from Lemma 3.4 that under the assumptions of Lemma 3.1, the sequence of induced metrics $(m_n)_{n \in \mathbb{N}}$ on $\partial C(M_n)$ is bounded.

Lemma 3.10 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Let λ_n be the measured bending laminations on the boundary of the convex core of M_n and suppose that $\lambda_n \to \lambda_\infty$. Let λ_{\pm} be the respective restrictions of λ_∞ to the two components of the boundary $\partial C(M_n)$ of the convex core of M_n . Suppose that

- λ_{-} and λ_{+} fill *S*,
- each closed curve in the support of λ_- (respectively λ_+) has weight less than π .

Then the sequence of induced metrics $(m_n)_{n \in \mathbb{N}}$ on $\partial C(M_n)$ is bounded.

Proof If (m_n) is unbounded, then there is a simple closed curve $d \subset S$ with geodesic representative $d_n \subset \partial C(M_n)$ such that $l_{m_n}(d_n)$ is unbounded. By Lemma 3.4 and the assumptions on λ , there is a sequence E_n of essential annuli such that $i(\lambda_n, \partial E_n)$ tends to 0. Such a sequence of annuli contradicts the assumption that λ_- and λ_+ fill S.

3.5 Convergence of convex cores

The last step in the proof of Lemma 3.1 is to show that, under the assumption that the sequence of metrics on the boundary are bounded, a subsequence of convex cores converges for the bilipschitz topology. Before starting the proof, we will discuss the Margulis lemma for quasifuchsian manifolds with particles. Let us first review the Margulis lemma for manifolds with variable curvature.

Theorem 3.11 (Margulis lemma) Given $n \in \mathbb{N}$ there are constant $\mu = \mu(n) > 0$ and $I(n) \in \mathbb{N}$ with the following property. Let X be an *n*-dimensional Hadamard manifold that satisfies the curvature condition $-1 \le K \le 0$ and let Γ be a discrete group of isometries acting on X. For $x \in X$, let $\Gamma_{\mu}(x) = \{\gamma \in \Gamma \mid d_{\gamma}(x) \le \mu\}$ be the subgroup generated by the elements γ with $d_{\gamma}(x) \le \mu$. Then $\Gamma_{\mu}(x)$ is almost nilpotent, thus it contains a nilpotent subgroup of finite index. The index is bounded in I(n).

This statement is taken from Ballmann, Gromov and Schroeder [4, Section 8.3]. Since we are considering a manifold M homeomorphic to $S \times I$, an almost nilpotent subgroup of $\pi_1(M)$ is cyclic.

Theorem 3.11 does not hold for hyperbolic manifolds with cone singularities since the curvature is not defined at the singularities. On the other it is not hard to replace the metric in a neighborhood of the singular locus with a Riemannian metric. Furthermore, if the cone angle is at least 2π , one can choose the Riemannian metric so that it has negative curvature with a lower bound depending on the cone angles and the choice of the neighborhood of the singularities. Now we consider a quasifuchsian manifold with particles M and a negatively curved branched cover \overline{M} of M. By Lemma 2.8, there are R, $\varepsilon > 0$ such that any closed curve with length at most ε is at distance at least R from the singularities. We replace the R-neighborhood of the singularities with a smooth Riemannian metric and apply Theorem 3.11 to the resulting manifold. Notice that the lower bound on the curvature of the Riemannian metric thus obtained will depend on R and the cone angle. Thus we get ε depending on R and the cone angles so that for a given point x in the universal cover \widetilde{M} of \overline{M} the subgroup of $\pi_1(\overline{M})$ generated by the set $\{\gamma \in \pi_1(\overline{M}) \mid d(x, \gamma x) \leq \varepsilon\}$ is cyclic. Since \overline{M} is a finite branched cover, we have a similar statement for M, replacing ε with ε/p , where p is

the index of the cover, which depends on the cone angles of the singularities of M. It follows that we have a Margulis decomposition for quasifuchsian manifolds with particles:

Lemma 3.12 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. There is an ε depending on the limit angles such that, for *n* large enough, each component of the ε -thin part of M_n is a neighborhood of a closed geodesic.

Notice that the Margulis tubes we obtain here are disjoint from the singularities so they are isometric to regular neighborhoods of geodesics in hyperbolic 3–manifolds. This result also follows from [10, Theorem 5.3]. We can now discuss the convergence of quasifuchsian manifolds with particles.

Lemma 3.13 Let M_n be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Suppose that the sequence $(m_n)_{n \in \mathbb{N}}$ (the induced metrics on the boundary of the convex cores) converges. Then, after taking a subsequence, $(M_n)_{n \in \mathbb{N}}$ converges to a quasifuchsian manifold with particles with the same topological type as M_n .

Proof First notice that since the cone angles are less than π , by Lemma 2.8 there is a positive lower bound for the distance between two components of the singularity locus. Consider a point $x_n \in C(M_n)$, extract a subsequence such that the sequence (x_n, M_n) converges in the Gromov–Hausdorff topology (such a subsequence always exists). By [11, Proposition 3.2.6], the limit (x_{∞}, M_{∞}) is a hyperbolic manifold with cone singularities. By [11, Proposition 3.3.1], the sequence (x_n, M_n) converges to (x_{∞}, M_{∞}) in the bilipschitz topology.

It remains to show that M_{∞} has the same topological type as M_n and that its metric is convex cocompact. To do that we will show that the diameter of $C(M_n)$ is uniformly bounded. It will follow that $C(M_n)$ converges to a convex set with the same topological type.

Lemma 3.14 Let $(M_n)_{n \in \mathbb{N}}$ be a sequence of quasifuchsian manifolds with particles with the same topological type and converging angles. Suppose that the sequence of induced metrics $(m_n)_{n \in \mathbb{N}}$ on the boundary of the convex cores converges. Then the diameter of $C(M_n)$ is uniformly bounded.

Proof Consider a negatively curved ramified cover \overline{M}_n of M_n whose topological type does not eventually depend on n. It follows from the Margulis lemma that a very short geodesic in \overline{M}_n lies in a very deep embedded tube. Using this observation we will show that there is a uniform lower bound on the length of any fixed curve in $C(\overline{M}_n)$.

Claim 3.15 Let M_n be a sequence of quasifuchsian manifolds with particles and let \overline{M}_n be a negatively curved ramified cover of M_n whose topological type does not depend on n. Suppose that the sequence $(\overline{m}_n)_{n \in \mathbb{N}}$ converges. Let $\overline{c} \subset \overline{S}$ be a simple closed curve. Then there is Q > 0 such that if $\overline{c}_n \subset C(\overline{M}_n)$ denotes the geodesic representative of \overline{c} , $l_{\overline{g}_n}(\overline{c}_n) \ge Q$ for any $n \in \mathbb{N}$.

Proof Assume the contrary, that is (after extracting a subsequence), $\lim l_{\overline{g}_n}(\overline{c}_n) = 0$. Then \bar{c}_n is the core of a deep Margulis tube \overline{T}_n . Notice that since $(\bar{m}_n^+, \bar{m}_n^-)_{n \in \mathbb{N}}$ converges, there is no short curve in $\partial C(M_n)$. More precisely, there is a uniform lower bound on the length of simple closed geodesics on $\partial C(M_n)$. Since the induced metric on $\partial C(\overline{M}_n)$ is negatively curved, it can have a large diameter only if it contains a short curve. Thus the uniform lower bound on the length of simple closed geodesics on $\partial C(\overline{M}_n)$ provides us with a bound on the diameter of each component of $\partial C(\overline{M}_n)$. It follows that $\partial C(M_n)$ does not go too deep into a Margulis tube (compare with Minsky [42, Lemma 6.3]): let ε_0 be a Margulis constant for the sequence M_n as provided by Lemma 3.12, namely the ε_0 -thin part $M_n^{<\varepsilon_0}$ of M_n is a union of Margulis tubes for *n* large enough. By Meyerhoff [41] and Brooks and Matelski [21] (see also [42, Lemma 6.1]), given ε small enough, the diameter of $M_n^{<\varepsilon_0} - M_n^{<\varepsilon}$ is large. In particular, if a component of $\partial C(\overline{M}_n)$ intersects $M_n^{<\varepsilon}$ for a small ε , it has a large diameter. Hence the bound on the diameter of each component of $\partial C(\overline{M}_n)$ provides us with a constant ε (depending on the sequence M_n) so that $\partial C(\overline{M}_n)$ is disjoint from the thin part $M_n^{<\varepsilon}$.

If we take \overline{T}_n to be an ε -Margulis tube, we get that \overline{T}_n lies entirely in the interior of $C(\overline{M}_n)$. Consider a simple closed curve $\overline{d} \subset \overline{S}$ that intersects \overline{c} essentially. By Lemma 3.2, there is an essential annulus $\overline{A}_n \subset C(\overline{M}_n)$, which is in the homotopy class defined by $\overline{d} \times I$ such that the area of \overline{A}_n is at most $\ell_{\overline{m}_n^+}(\overline{d}_+) + \ell_{\overline{m}_n^-}(\overline{d}_-)$. In particular, since the sequence $(\overline{m}_n)_{n \in \mathbb{N}}$ converges, the area of \overline{A}_n is bounded. On the other hand, since \overline{d} intersects \overline{c} essentially, \overline{A}_n intersects \overline{c}_n essentially. In particular, \overline{A}_n intersects \overline{T}_n along a disc \overline{D}_n . When the length of \overline{c}_n tends to 0, $d(\overline{c}_n, \partial \overline{T}_n) \to \infty$ (see [21; 41; 42, Lemma 6.1]). It follows that the diameter of \overline{D}_n , and hence its area, tends to ∞ when the length of \overline{c}_n tends to 0. Thus an upper bound for the area of $\overline{A}_n \supset \overline{D}_n$ yields a lower bound for the length of \overline{c}_n . This concludes the proof of Claim 3.15. Consider now two simple closed curves $\bar{c}, \bar{d} \subset \bar{S}$ such that the components of $\bar{S} \setminus (\bar{c} \cup \bar{d})$ are discs. Two such curves are said to fill the surface \bar{S} . Consider essential annuli \bar{A}_n and \bar{B}_n in $C(\bar{M}_n)$ in the homotopy classes defined by \bar{c} and \bar{d} , constructed as in Lemma 3.2. In particular \bar{A}_n and \bar{B}_n have bounded area. Since \bar{A}_n and \bar{B}_n have bounded areas and negative curvature, the only way for them to have a large diameter is to have a very short core curve. But this would contradict Claim 3.15. Thus we can conclude that \bar{A}_n and \bar{B}_n have uniformly bounded diameters.

Let $\mathcal{B}_{1 \leq k \leq p}$ be the closure of the components of $\overline{S} \times I \setminus (\overline{c} \times I \cup \overline{d} \times I)$. Our manifold $\overline{N} = \overline{S} \times I$ is the union of the \mathcal{B}_k and the \mathcal{B}_k are all balls. Define a surjective map $f_n: \overline{S} \times I \to C(\overline{M}_n)$ that maps $\overline{c} \times I$ and $\overline{d} \times I$ to \overline{A}_n and \overline{B}_n respectively and such that the restriction of f_n to each \mathcal{B}_k is an immersion. For each k, the image of $\partial \mathcal{B}_k$ lies in $\overline{A}_n \cup \overline{B}_n \cup \partial C(M_n)$. Since \overline{A}_n and \overline{B}_n have bounded diameters and since the induced metric on $\partial C(\overline{M}_n)$ is bounded, the diameter of $f_n(\partial \mathcal{B}_k)$ is bounded for any k. It follows that $f_n(\mathcal{B}_k)$ has a bounded diameter for any k. Since f_n is surjective, this implies that $C(\overline{M}_n)$ has a uniformly bounded diameter. Since the index of the cover $\overline{M}_n \to M_n$ does not depend on n, $C(M_n)$ has a bounded diameter. \Box

It remains to show that the convex core of M_{∞} is compact and homeomorphic to $S \times I$ (when M_n is homeomorphic to $S \times \mathbb{R}$). Once again we will use the negatively curved ramified cover \overline{M}_n . Since $(\overline{x}_n, \overline{M}_n)$ converges to $(\overline{x}_\infty, \overline{M}_\infty)$, there is a sequence $R_n \to \infty$ and a sequence of bilipschitz maps $\phi_n: B(\bar{x}_n, R_n) \to B(\bar{x}_\infty, R_n)$ such that the bilipschitz constants tend to 1. By Lemma 3.14, for *n* large enough, $C(\overline{M}_n) \subset B(x_n, R_n)$. Given a geodesic segment $\gamma_n \subset C(\overline{M}_n), \phi_n(\gamma_n)$ almost realizes the distance between its endpoints. Since \overline{M}_n is a hyperbolic manifold with cone singularities, $\phi_n(\gamma_n)$ is very close to the geodesic segment joining its endpoints. It follows that, for n large enough, the convex hull of $\phi_n(C(\overline{M}_n))$ lies in a small neighborhood $\mathcal{V}_n(\phi_n(C(\overline{M}_n)))$ of $\phi_n(C(\overline{M}_n))$. This convex hull has to contain $C(\overline{M}_\infty)$ since it is the smallest convex set. Thus we have $C(\overline{M}_{\infty}) \subset \mathcal{V}_n(\phi_n(C(\overline{M}_n)))$. It follows that $C(\overline{M}_{\infty})$ is compact. Furthermore, since the induced metric on $\partial C(\overline{M}_n)$ is bounded, $\mathcal{V}_n(\phi_n(C(\overline{M}_n)))$ is homeomorphic to $\overline{S} \times I$ for *n* large enough. It follows that $C(\overline{M}_\infty)$ is homeomorphic to $\overline{S} \times I$. Thus we have proved that M_{∞} is a quasifuction manifold with cone singularities with the same topological type as M_n .

In contrast to the other results of this section, we do not need to assume that the quasifuchsian manifolds under consideration in Lemma 3.13 are not fuchsian.

3.6 The bending lamination of the convex core

To finish the proof of Lemma 3.1 we only have to check that the induced bending lamination on the boundary of the convex core of the limit manifold is the limit of the bending laminations. We can state the result as follows.

Lemma 3.16 Let $N = S \times \mathbb{R}$, let x_1, \ldots, x_{n_0} be distinct points on S, and let $\kappa_i = \{x_i\} \times \mathbb{R}, 1 \le i \le k$. Let $(g_n)_{n \in \mathbb{N}}$ be a sequence of quasifuchsian metrics on N with particles of angles θ_n^i along κ_i , $1 \le i \le k$. Let λ_n , respectively m_n , be the measured bending laminations, respectively the induced metric, on the boundary of the convex core of (N, g_n) . Suppose that (g_n) converges in bilipschitz topology towards a quasifuchsian metric with particles g on N, with cone angles $\theta^i \in (0, \pi)$ along κ_i . Then $(m_n)_{n \in \mathbb{N}}$ converges to the induced metric m on the boundary of the convex core of (N, g), while $(\lambda_n)_{n \in \mathbb{N}}$ converges to the measured bending lamination λ of the boundary of the convex core of (N, g).

Proof Set M = (N, g) and $M_n = (N, g_n)$ and denote by C(M) the convex core of M and by $C(M_n)$ the convex core of M_n . We consider as above the finite cover \overline{N} of N ramified along the cone singularities, chosen so that all cone angles in \overline{N} have angle larger than 2π . This is useful below since we will use negative curvature arguments, in particular the existence of a geodesic segment in a homotopy class with fixed endpoints. Clearly it is sufficient to prove the lemma for \overline{N} , where the "convex core" considered is $C(\overline{M}_n)$, the lift to \overline{N} of $C(M_n)$, since once the result is obtained in \overline{N} , we can take the quotient by the group of deck transformations to obtain the result on N.

Let $(\gamma_n)_{n \in \mathbb{N}}$ be a sequence of segments in \overline{N} , with γ_n geodesic for \overline{g}_n for all $n \in \mathbb{N}$. Suppose that $(\gamma_n)_{n \in \mathbb{N}}$ converges to a segment γ . We know that $\overline{g}_n \to \overline{g}$ in the bilipschitz topology and, in hyperbolic geometry, any segment which is close to realizing the distance between its endpoints is close to a geodesic segment. So γ is geodesic for \overline{g} . Conversely, any geodesic segment for \overline{g} is a Hausdorff limit of geodesic segments for the g_n . The same holds for closed geodesics.

Let $(\Omega_n)_{n \in \mathbb{N}}$ be a sequence of compact subsets of \overline{N} such that, for all $n \in \mathbb{N}$, Ω_n is convex for \overline{g}_n . Suppose that $\Omega_n \to \Omega$ in the Hausdorff topology. The definition of a convex subset and the previous paragraph show that Ω is convex, since any geodesic segment γ in $(\overline{N}, \overline{g})$ with endpoints in the interior of Ω is the limit of a sequence of geodesic segments γ_n , with γ_n geodesic for \overline{g}_n . Since γ_n has endpoints in Ω_n (for *n* large enough) and Ω_n is convex for \overline{g}_n , $\gamma_n \subset \Omega_n$, and therefore $\overline{\gamma} \subset \Omega$, and Ω is convex for \overline{g} . Conversely, a similar argument shows that any compact convex subset for \overline{g} is the Hausdorff limit of a sequence of compact convex subsets of the metrics \overline{g}_n .

For all n, $C(\overline{M}_n)$ contains all closed geodesics in $(\overline{N}, \overline{g}_n)$. Given a nontrivial homotopy class α in \overline{N}_r (the complement of the singular curves in \overline{N}), it is realized for each $n \in \mathbb{N}$ by a (unique) closed geodesic γ_n in $(\overline{N}, \overline{g}_n)$, and the sequence $(\gamma_n)_{n \in \mathbb{N}}$ converges to the closed geodesic γ , which realizes α in $(\overline{N}, \overline{g})$. For each $n \in \mathbb{N}$, $\gamma_n \subset C(\overline{M}_n)$. Moreover we have seen that the diameter of the $C(\overline{M}_n)$ is bounded. It follows that $(C(\overline{M}_n))_{n \in \mathbb{N}}$ converges — after extracting a subsequence — to a limit subset C', which contains all closed geodesics in $(\overline{N}, \overline{g})$.

Since C' is the limit of a sequence of convex subset of the $(\overline{N}, \overline{g}_n)$, it is convex. Moreover if $\Omega \subset C'$ is convex, then it is the limit of a sequence of convex subsets $\Omega_n \subset C(\overline{M}_n)$. But then $(\Omega_n \cap C(\overline{M}_n))_{n \in \mathbb{N}}$ is a sequence of convex subsets converging to Ω . Because the $C(\overline{M}_n)$ are minimal convex subsets, $\Omega_n \cap C(\overline{M}_n) = C(\overline{M}_n)$ for all n, so that $\Omega = C'$. So $C' = C(\overline{M})$. This shows that $C(\overline{M})$ is the Hausdorff limit of the $C(\overline{M}_n)$.

Note that it is not clear at this point whether $\partial C(\overline{M}_n) \to \partial C(\overline{M})$ in the C^1 topology. However, a general fact is that, if $\phi: S \to H^3$ is a smooth embedding of a surface, and if $(\phi_n)_{n \in \mathbb{N}}$ is a sequence of Lipschitz embeddings of S in H^3 which converges to ϕ in the C^0 topology, then the distance d_n induced on S by the ϕ_n are larger in the limit than the distance d induced by ϕ :

(1)
$$\limsup_{n \to \infty} d_n(x, y) \ge d(x, y) \quad \text{for all } x, y \in S.$$

The same holds if ϕ is Lipschitz with locally convex image rather than smooth; see Aleksandrov and Zalgaller [2]. Moreover, in case of equality in Equation (1), and if the ϕ_n also have locally convex images, then the convergence of (ϕ_n) to ϕ is stronger, in the sense that the tangent plane to $\phi_n(S)$ converges almost everywhere to the tangent plane to $\phi(S)$.

Coming back to $\partial C(\overline{M}_n)$, the C^0 convergence towards $\partial C(\overline{M})$ (together with the bilipschitz convergence of \overline{g}_n to \overline{g}) is sufficient to insure that the metric \overline{m}_n on $\partial C(M_n)$ is *larger* in the limit than the metric \overline{m} induced by \overline{g} on $\partial C(\overline{M})$. In other terms, if $x, y \in \partial C(\overline{M})$ and $x_n, y_n \in \partial C(\overline{M}_n)$ are such that $\lim x_n = x, \lim y_n = y$, then there exists for each $\epsilon > 0$ some $N_0 \in \mathbb{N}$ such that, for all $n \ge N_0$,

$$d_{\overline{m}_n}(x_n, y_n) \ge (1 - \epsilon) d_{\overline{m}}(x, y).$$

It follows that the lengths of the closed geodesics in S for \overline{m}_n are bounded from below by $(1 - \epsilon)$ times their lengths for \overline{m} .

Since the metrics \overline{m}_n and \overline{m} are hyperbolic metrics with cone singularities of fixed angles, this shows, using standard arguments based for instance on pants decompositions, that $\overline{m}_n \to \overline{m}$ (see Section A.2 or [24]).

It then follows that as $n \to \infty$, \overline{m}_n is bounded from above by $(1 + \epsilon)\overline{m}$: under the same hypothesis as above,

(2)
$$d_{\overline{m}_n}(x_n, y_n) \le (1+\epsilon)d_{\overline{m}}(x, y).$$

Since $\overline{m}_n \to \overline{m}$ in this sense of Equation (2), and moreover $\partial C(\overline{M}_n) \to \partial C(\overline{M})$ in the C^0 topology, it follows that the convergence is actually stronger, and the tangent plane to $\partial C(\overline{M}_n)$ converges almost everywhere to the tangent plane to $\partial C(\overline{M})$ (both exist almost everywhere by convexity). This implies that the measured laminations λ_n of ∂C_n converge to λ .

4 Prescribing the measured bending lamination on the boundary of the convex core

The goal of this section is to prove Theorem 1.12 and then Theorem 1.13. The proof of Theorem 1.12 is largely based on a well-known doubling argument already used for nonsingular manifolds, which reduces the infinitesimal rigidity with respect to the measured lamination (when the support of the lamination is along closed curves) to a rigidity statement proved by Hodgson and Kerckhoff [29] for hyperbolic conemanifolds.

Theorem 1.13 is then a consequence, using the compactness statement proved in Section 3.

4.1 A doubling argument

Let M be convex cocompact manifold with particles, and let C(M) be its convex core. Suppose that the support of the measured bending lamination of C(M) is a disjoint union of closed curves.

Definition 4.1 The *doubled convex core* of M is the 3-dimensional hyperbolic manifold with cone singularities DC(M) obtained by gluing two copies of C(M) isometrically using the identification of their boundaries.

We have seen that the singular locus of M does not intersect the support of the bending lamination on the boundary of the convex core; actually it even remains at a distance

that is bounded from below by a constant depending only on the cone angles. So the "particles" intersect the boundary of the convex core inside faces, and moreover they do so orthogonally. It follows that the singular locus of DC(M) is a disjoint union of closed curves, which are of two types:

- Each "particle" p of M corresponds to a cone singularity along a closed curve in DC(M), of length equal to twice the length of the intersection of p with C(M).
- Each closed curve in the support of the measured bending lamination of the boundary of C(M) corresponds to a closed curve (of the same length) in DC(M).

Still by definition, DC(M) admits an isometric involution—exchanging the two copies of C(M) which are glued to obtain DC(M)—and the set of fixed points of this involution is a (not connected) closed surface S, which corresponds to the boundaries of both copies of C(M). This surface is orthogonal to the singularities of the first kind, and contains the singularities of the second kind.

4.2 Local deformations

The doubling trick explained above leads directly to a rigidity statement. We consider again a convex cocompact manifold M with particles, for which the measured bending lamination of the convex core is along closed curves $\gamma_1, \ldots, \gamma_N$, for which the bending angles are equal to $\alpha_1, \ldots, \alpha_N \in (0, \pi)$. As in the introduction, we call $\theta_1, \ldots, \theta_{n_0}$ the cone angles at the "particles", and let $\theta = (\theta_1, \ldots, \theta_{n_0})$.

Lemma 4.2 There exists a neighborhood U of $(\alpha_1, \ldots, \alpha_N)$ in $(0, \pi)^N$ and a neighborhood V of the hyperbolic metric g on M in $\mathcal{QF}_{S,n_0,\theta}$ such that, if $(\alpha'_1, \ldots, \alpha'_N)$ lies in U, there is a unique $g' \in V$ for which the support of the measured bending lamination on C(M) is $\gamma_1 \cup \cdots \cup \gamma_N$ and the bending angle on γ_i is α'_i , $1 \le i \le N$.

Proof Hodgson and Kerckhoff [29] proved a local deformation result for hyperbolic cone-manifolds. It follows from their result that there exists a unique cone-manifold close to DC(M) with the same topology as DC(M) (including the singular locus), the same angles at the cone singularities corresponding to the particles in M, and angles $2\alpha'_1, \ldots, 2\alpha'_N$ instead of $2\alpha_1, \ldots, 2\alpha_N$ at the cone singularities corresponding to the pleating lines of C(M).

The uniqueness of D' shows that it has the same symmetry as DC(M), that is, it admits an isometric involution fixing a surface S' isotopic to the surface S fixed

by the isometric involution on DC(M). By an easy symmetry argument, the cone singularities in D' corresponding to the particles in M still have to be orthogonal to S', while those corresponding to the pleating lines of $\partial C(M)$ have to be contained in S' (see [12, Section 8] for details on the uniqueness part of this argument; the same argument can basically be used when particles are present).

Therefore, D' is the double of a hyperbolic manifold with convex boundary (obtained as the metric completion of one half of the complement of S' in D') with cone singularities orthogonal to the boundary. The boundary of this manifold is convex with no extremal point, so that it is the convex core of a quasifuchsian manifold with particles M', with the same cone angle as M at the particles and such that $\partial C(M')$ is pleated along the same lines as $\partial C(M)$, but with pleating angles $\alpha'_1, \ldots, \alpha'_N$ instead of $\alpha_1, \ldots, \alpha_N$.

The uniqueness of such a manifold, in the neighborhood of M, follows from the uniqueness of D' in the neighborhood of DC(M).

4.3 Proof of Theorem 1.12

Let $\gamma_1, \ldots, \gamma_N$ be the curves in the support of λ , considered as curves in ∂N . Following the doubling construction above, we define a closed manifold D(N) by gluing two copies of N along their boundary. D(N) contains two families of curves, which we still call c_1, \ldots, c_{n_0} (corresponding to the particles in N) and $\gamma_1, \ldots, \gamma_N$ (corresponding to the pleating lines on the boundary of the convex core).

Let $\theta'_1, \ldots, \theta'_{n_0} \in (0, \pi)$ and $\alpha'_1, \ldots, \alpha'_N \in (0, \pi)$ be chosen such that:

- For all $i \in \{1, \ldots, n_0\}$, $0 \le \theta'_i \le \theta_i$, and $\theta'_i = \pi/k_i$ for some $k_i \in \mathbb{N}$.
- For all $j \in \{1, \ldots, N\}$, $0 \le \alpha'_j \le \alpha_j/2$, and $\alpha'_j = \pi/2l_j$ for some $l_j \in \mathbb{N}$.

The orbifold hyperbolization theorem for cyclic orbifolds (initially stated by Thurston, and proved in [11; 23]) can be applied to show that there is a unique hyperbolic orbifold structure on D(N) with singularities of angles θ'_i on the c_i and $2\alpha'_j$ on the γ_j . Since it is unique, it has an isometric involution exchanging the two copies of N glued to obtain D(N) and thus is the double of the convex core of a quasifuchsian manifold with particles.

Let $(\theta_t)_{t \in [0,1]} = (\theta_{1,t}, \dots, \theta_{n_0,t})_{t \in [0,1]}$ and $(\alpha_t)_{t \in [0,1]} = (\alpha_{1,t}, \dots, \alpha_{N,t})_{t \in [0,1]}$ be the 1-parameter families defined by

$$\theta_{i,t} = (1-t)\theta'_i + t\theta_i, \quad 1 \le j \le n_0,$$

$$\alpha_{j,t} = (1-t)\alpha'_j + t\alpha_j, \quad 1 \le j \le N.$$

Geometry & Topology, Volume 18 (2014)

Let $I \in [0, 1]$ be the maximal interval containing 0 such that, for all $t \in I$:

- There exists a hyperbolic structure on D(N) with cone singularities of angle $\theta_{i,t}$ on c_i , $1 \le i \le n_0$, and a cone singularity of angle $2\alpha_{j,t}$ on γ_j , $1 \le j \le N$.
- This hyperbolic structure has an isometric involution exchanging the two copies of N glued to obtain D(N).

By construction, $I \neq \emptyset$. Lemma 4.2 shows that *I* is open, while Lemma 3.1 shows that *I* is closed. So I = [0, 1]; this proves the existence part of the statement because D(N), with the hyperbolic cone-structure for t = 1, is obtained by doubling the convex core of a convex cocompact hyperbolic manifold with particles of angles θ_i and pleating angles α_i on the boundary, as needed.

For the uniqueness, the same deformation argument can be used to start from a conemanifold structure on D(N) and decrease the angles along the curves γ_j , $1 \le j \le N$, from $2\alpha_j$ to $2\alpha'_j$. Lemma 4.2 shows that the corresponding deformation of the hyperbolic cone-manifold structure exists and is unique. Since the endpoint of the deformation is unique (by the Orbifold Hyperbolization Theorem) there can be only one cone-manifold structure on D(N) with angles θ_i on the curves c_i , $1 \le i \le n_0$, angle $2\alpha_j$ on the curve γ_j , $1 \le j \le N$, and the necessary symmetry property.

4.4 Proof of Theorem 1.13

Given $\lambda_{-}, \lambda_{+} \in \mathcal{ML}_{S,x}$ satisfying the hypothesis of Theorem 1.13, both are limits of a sequence of measured laminations $(\lambda_{-,n})_{n \in \mathbb{N}}, (\lambda_{+,n})_{n \in \mathbb{N}}$ with support along a union of closed curves, which satisfy the hypothesis of Theorem 1.12.

For all *n*, Theorem 1.12 shows that $\lambda_{-,n}$ and $\lambda_{+,n}$ are the upper and lower measured bending laminations of the boundary of the convex core for a unique quasifuchsian hyperbolic structure with particles g_n on $S \times \mathbb{R}$. Lemma 3.1, applied to this sequence of hyperbolic structures, shows that it has a subsequence that converges to a quasifuchsian hyperbolic structure with particles, for which the lower and upper measured bending laminations of the boundary of the convex core are λ_- and λ_+ , respectively.

4.5 The conditions are necessary

Finally we check here that the hypothesis in Theorem 1.13 are necessary. It obviously follows that the hypothesis in Theorem 1.12 are also necessary.

Lemma 4.3 Let *M* be a non-Fuchsian quasifuchsian manifold with particles, let λ be the measured bending lamination on the boundary of its convex core. Then λ satisfies the hypothesis of Theorem 1.13.

Proof The hypothesis that the weights of each closed curve in the support of λ_{-} and λ_{+} is less than π is clearly a consequence of the fact that C(M) is convex and compact.

Suppose by contradiction that λ_{-} and λ_{+} do not fill *S*. There exists then a sequence $(c_n)_{n \in \mathbb{N}}$ of simple closed curve in *S* such that

$$i(\lambda_{-}, c_n) + i(\lambda_{+}, c_n) \to 0$$

Let c_n^- and c_n^+ be the geodesic representatives of c_n in the lower and upper boundary components of C(M), respectively.

Let \bar{c}_n^- and \bar{c}_n^+ be lifts of c_n^- and c_n^+ , respectively, to \bar{M} , corresponding to the same lift of c. Lemma 3.2 shows that there exists an annulus $A_n \subset \bar{M}$ bounded by \bar{c}_n^- and \bar{c}_n^+ on which the induced metric is hyperbolic with cone points of negative singular curvature (cone angle larger than 2π). Moreover, the boundary of A_n is convex (for the induced metric) and its total curvature goes to 0 as $n \to \infty$. By Lemma 3.2 the area of A_n goes to 0 as $n \to \infty$. Since the lengths of the \bar{c}_n^- and \bar{c}_n^+ are bounded from below, this means that the distance between \bar{c}_n^- and \bar{c}_n^+ in A_n goes to 0 as $n \to \infty$. Therefore, the distance between the upper and lower boundary of C(M) is zero, a contradiction because we have supposed that M is not Fuchsian. \Box

5 Earthquakes estimates

In this section we consider a convex cocompact manifold with particles M. The arguments in this more general case are the same as in the specific situation of quasifuchsian manifolds with particles. Its boundary ∂M has a number of marked points x_1, \ldots, x_{2n_0} , which are the endpoints of the n_0 "particles", and to each is attached an angle $\theta_k \in (0, \pi)$, $1 \le k \le 2n_0$, which is the angle at the corresponding particle.

We identify ∂M with the boundary of its convex core (see Lemma 2.5). We will use the following notation.

- λ is the measured bending lamination of the boundary of the convex core.
- *m* is its induced metric.
- t is the (unique) hyperbolic metric in the conformal class at infinity τ, with cone singularities of prescribed angle θ_k at the marked point x_k.
- G_λ(m) is the metric obtained by grafting the hyperbolic metric m along the measured lamination λ, so that G_λ(m) has curvature in [-1, 0]. If for instance λ is rational, then G_λ(m) is obtained by inserting a flat annulus in (∂M, m) for each closed curve in the support of λ; see Dumas [25].

This section contains a basic estimate relating t to m. It will be useful in proving the compactness of a certain map and Theorem 1.7. Its statement is based on the following extension to hyperbolic surfaces with cone singularities of Thurston's earthquake theorem (as found in [32; 55]).

Theorem 5.1 [13] For any $h, h' \in \mathcal{T}_{\Sigma,x,\theta}$, there is a unique measured lamination $\nu \in \mathcal{ML}_{\Sigma,x}$ such that the right earthquake along ν sends h to h'.

The main estimate proved in this section, and the main tool for the proof of Theorems 1.7 and 1.14, is the following.

Proposition 5.2 There exists a constant C > 0 (depending only on the topology of M) such that, if $v \in \mathcal{ML}_{\partial M,x}$ is the measured lamination such that $t = E_r(v)(m)$, then the length $L_m(v)$ is at most C.

This proposition is proved in Section 5.4, after some preliminary considerations. It is used below in Section 6.

5.1 The average curvature of geodesics

In this part we prove a technical statement that is useful at several points below. It is an extension to convex cocompact manifolds with particles of a result proved earlier by Bridgeman [15] for the convex core of nonsingular convex cocompact manifolds, or more generally of pleated surfaces in H^3 . However the argument used here is inspired by Bonahon and Otal [12].

We consider a quasifuchsian manifold with particles M and call $\theta_1, \ldots, \theta_{n_0}$ the cone angles at the particles. By definition, $\theta_1, \ldots, \theta_{n_0} \in (0, \pi)$. Here S is one of the connected components of $\partial C(M)$.

Proposition 5.3 There exists a constant $C_0 > 0$ such that if γ is a geodesic segment on *S* transverse to λ , $i(\gamma, \lambda) \leq C_0(l_m(\gamma) + 1)$.

Note that C_0 depends on the θ_i (at least the argument we use here does depend on the maximum of the θ_i) but not otherwise on M.

Proposition 5.3 will follow from the following lemma. We use here the constant ϵ_0 coming from Lemmas 2.7 and 2.8.

Lemma 5.4 There exists $\lambda_1 > 0$ such that if γ is at distance at least $\epsilon_0/2$ from the intersection of *S* with the singular set of *M* and if $l_m(\gamma) \le \epsilon_0/4$, then $i(\lambda, \gamma) \le \lambda_1$.

The proof of this lemma is based on some intermediate steps. The first is a consequence of Lemmas 2.7 and 2.8.

Claim 5.5 There exists $\rho_0 > 0$ such that any point in *S* at distance at least ϵ_0 in *S* from the singular points of *S* is also at distance at least ρ_0 in *M* from the singular set of *M*.

Proof Let $x \in S$, which is at distance at least ϵ_0 from the singular points of *S*; suppose that it is at distance strictly less than ϵ_0 from a particle *p*. Let *y* be a point in *p* that is closest to *x*, and let *D* be the totally geodesic disk of radius ϵ_0 orthogonal to *p* at *y*. This disk does not encounter any other particle by Lemma 2.8. Moreover $x \in D$ because *y* is at minimal distance from *x* among the points of *p*.

Let Ω equal to the intersection of D with the convex core C(M) of M. Notice that $\partial D \cap \Omega \neq \emptyset$ for otherwise we would have $\partial \Omega \subset \partial C(M)$ and $\partial \Omega$ would bound a disc Ω' in $\partial C(M)$. Since D is totally geodesic and C(M) is convex, there would be a point $x \in \Omega'$ such that the link of C(M) at x is contained in an open hemisphere. This is impossible since $\partial C(M)$ is ruled.

We can therefore apply the second point in Lemma 2.7 to D, with Ω as defined above, ie equal to the intersection of D with the convex core of M. The result follows. \Box

Corollary 5.6 Let $x \in S$ be contained in the support of the bending lamination λ , and let D' be the totally geodesic disk of radius ρ_0 in M orthogonal to λ at x. Then D' does not intersect the singular set of M.

After taking ρ_0 smaller if necessary, we have another simple statement which will be necessary below.

Claim 5.7 Let $y \in \lambda$ be a point in the connected component of x in the intersection of S with D', and let g_y be the geodesic segment in the support of λ centered at y and of length $2\epsilon_0$. Then the angle between g_y and D' at y is at least $\pi/4$.

Proof We call g_x the geodesic segment contained in the support of λ centered at x and of length $2\epsilon_0$. g_x is disjoint from g_y on S while x is at distance at most ρ_0 from y; it follows that there exists c > 0 (depending on ϵ_0 and ρ_0 , and going to 0 as $\rho_0 \rightarrow 0$ for fixed ϵ_0) such that the distance to g_x in S of any point of $g_y([-\epsilon_0 + \rho_0, \epsilon_0 - \rho_0])$ is at most $c\epsilon_0$. The same estimate holds in \tilde{M} , the universal cover of M. If ρ_0 is small enough—relative to ϵ_0 —the result follows.

Remark 5.8 There exists $k_0 > 0$, depending only on ρ_0 , such that, if Ω is a convex subset in the disk of radius ρ_0 in H^2 , the total curvature of the boundary of Ω is at most k_0 .

2348

Geometry & Topology, Volume 18 (2014)

Proof This follows from the Gauss–Bonnet theorem applied to Ω , with k_0 equal to 2π plus the area of the hyperbolic disk of radius ρ_0 .

Proof of Lemma 5.4 If γ does not intersect the support of λ the statement obviously holds, so we suppose that some point $x \in \gamma$ is in the support of λ . Let *D* be the totally geodesic disk of radius $\epsilon_0/2$ centered at *x* and orthogonal, at *x*, to the support of λ . By construction *D* is disjoint from M_s .

Remark 5.8 shows that the total curvature of the connected component c of $D \cap S$ containing x is at most k_0 . By Claim 5.7, each geodesic in the support of λ that intersects c makes an angle of at least $\pi/4$ with D. It follows that $i(c, \lambda) \leq 2k_0$. It also follows, since the length of γ is less than $\epsilon_0/4$, that γ can be deformed transversally to λ to a segment of c, so that $i(\gamma, \lambda) \leq i(c, \lambda)$. Therefore $i(\gamma, \lambda) \leq 2k_0$, and this proves the lemma.

Proof of Proposition 5.3 Notice first that Lemma 5.4, although stated only for geodesic segments γ that are at distance at least $\epsilon_0/2$ from the cone singularities, actually applies without this hypothesis. This is because, by Lemma 2.7, the support of λ cannot enter the ϵ_0 -neighborhood of the singular points, so that any part of γ at distance less than ϵ_0 from the singular set of S has zero intersection with λ .

Let $n \in \mathbb{N}$ be the unique integer such that $n\epsilon_0/4 \le l_m(\gamma) < (n+1)\epsilon_0/4$. Then γ can be cut into a sequence of segments $\gamma_1, \ldots, \gamma_n$ of length $\epsilon_0/4$ and one last segment γ_{n+1} of length smaller than $\epsilon_0/4$. Lemma 5.4 can be applied to each of those segments: it yields that $i(\lambda, \gamma_i) \le \lambda_1$, $1 \le i \le n+1$, so that

$$i(\lambda,\gamma) \leq (n+1)\lambda_1 \leq \left(\frac{4l_m(\gamma)}{\epsilon_0}+1\right)\lambda_1.$$

This proves the proposition.

5.2 The grafted metric and the hyperbolic metric at infinity

We consider here the relation between the grafted metric $G_{\lambda}(m)$ and the hyperbolic metric at infinity *t*.

We first recall the definition of the grafting map

$$G: \mathcal{ML} \times \mathcal{T} \to \mathcal{T},$$
$$(l, m) \mapsto G_l(m)$$

on a closed surface S.

Geometry & Topology, Volume 18 (2014)

The definition of $G_l(m)$ is simpler when l is a weighted multicurve, that is, its support is a disjoint union of closed curves c_1, \ldots, c_N . The transverse measure is then described by a positive weight w_i on c_i , for $1 \le i \le N$. Then $G_l(m)$ is obtained by realizing each curve c_i as a closed geodesics in (S, m), cutting S open along each c_i , and gluing in a flat strip of width w_i . Thurston showed that this map extends by continuity from weighted multicurves to measured laminations; see Kulkarni and Pinkall [36].

We now return to the setting where m and λ are the induced metric and measured bending lamination of the convex core of a quasifuchsian manifold M. The grafted metric $G_{\lambda}(m)$ is then isometric to the induced metric on the unit normal bundle of C(M) in the unit tangent bundle of M. (The unit normal bundle of C(M) is the space of unit vectors at points of $\partial C(M)$ that are the oriented normals of a support plane of C(M).)

One of the key properties of the grafted metric $G_{\lambda}(m)$ (see [36]) is that it is in the conformal class τ at infinity; more precisely, there is a natural "Gauss map" defined from the unit normal bundle of $\partial C(M)$ with its "grafted metric" to the boundary at infinity of M, which is conformal. This means that $G_{\lambda}(m)$ is conformal to t. Moreover, since the angles θ_i are in $(0, \pi)$, the intersection of the boundary of the convex core with the particles is at nonzero distance (for m) from the support of λ , so that the cone angles of the grafted metric at the intersections with the particles of the boundary of the convex core are equal to the cone angles of the corresponding singularities.

The fact that t is conformal to $G_{\lambda}(m)$ translates as $t = e^{2u}G_{\lambda}(m)$, where $u: \partial M \to \mathbb{R}$ is a function.

Lemma 5.9 The function u is nonpositive on ∂M .

Proof Consider two metrics g and g' with $g' = e^{2u}g$, and let K and K' be their curvatures. Then (see Besse [9, Chapter 1])

$$K' = e^{-2u} (\Delta u + K).$$

We can apply this formula here with $g = G_{\lambda}(m)$ and g' = t, so that K' = -1 while $K \in [-1, 0]$. It takes the form

$$\Delta u = -K - e^{2u} = |K| - e^{2u}$$

with $K \in [-1, 0]$ (this equation is understood in a distributional sense).

Since the cone angles are the same for t and for $G_{\lambda}(m)$, u is continuous and bounded at the singular points (see [56]). Let $x_M \in S$ be a point where u achieves its maximum. Suppose first that x_M is not a singular point. Then u is C^2 at x_M by elliptic regularity (see [56]). Moreover $\Delta u \ge 0$ at x_M since x_M is a maximum of u. It follows that $e^{2u} \le |K| \le 1$, so that $u \le 0$. To complete the proof it is sufficient to prove that u cannot achieve a positive maximum at a singular point of S. So we consider a singular point x_0 of S, and suppose that u > 0 at x_0 . We will show that u cannot have a maximum at x_0 .

Let *D* be the geodesic disk of radius *r* centered at x_0 in $(S, G_{\lambda}(m))$. Since λ does not enter a small neighborhood of x_0 , *D* is hyperbolic, with only one cone singularity at x_0 , if *r* is small enough. Let i_0 be the isometric map between *D*, with the metric $G_{\lambda}(m)$, and the hyperbolic disk H_{α}^2 with one cone singularity of angle α , where α is the cone angle of *S* at x_0 . Let $i_1: D \to H_{\alpha}^2$ be the isometric embedding of (D, t) in H_{α}^2 . Call v_0 the vertex of H_{α}^2 , ie its singular point. Since u > 0 at x_0 , if *r* is small enough, then

$$d(i_1(x), v_0) > d(i_0(x), v_0)$$
 for all $x \in D \setminus \{x_0\}$.

There is a natural complex map $\phi: H^2_{\alpha} \to H^2$, given in holomorphic coordinates centered at the singular point by $z \to z^{2\pi/\alpha}$. It is conformal and multiplies the metric by a factor $(2\pi/\alpha)^2 d(x, v_0)^{2(2\pi/\alpha-1)}$. Consider the composition

$$\Phi := \phi \circ i_1 \circ i_0^{-1} \circ \phi^{-1} \colon (\phi \circ i_0)(D) \to (\phi \circ i_1)^{-1}(D).$$

It is a conformal map, with conformal factor equal to

$$(2\pi/\alpha)^2 d(i_1(x), v_0)^{2(2\pi/\alpha-1)} e^{2v} (2\pi/\alpha)^{-2} d(i_0(x), v_0)^{-2(2\pi/\alpha-1)},$$

with $v = u \circ (\phi \circ i_0)^{-1}$. This can be written as

$$\left(\frac{d(i_1(x), v_0)}{d(i_0(x), v_0)}\right)^{2(2\pi/\alpha - 1)} e^{2v}$$

and is bigger than 1 since u > 0 and $d(i_1(x), v_0) > d(i_0(x), v_0)$.

Since Φ is a conformal map between two hyperbolic domains, its conformal factor cannot have a local maximum bigger than 1 at an interior point, by the argument given at the beginning of this proof. Therefore, *u* cannot have a positive maximum at x_0 . \Box

The following notion will be useful in this section and the next.

Definition 5.10 A *c*-*curve* on ∂M is either a closed curve or a segment with endpoints at cone singularities, which does not contain any singular point (except at its endpoints if it is not a closed curve).

We will sometimes implicitly consider c-curves up to homotopy in the complement of the singular points in ∂M . Each homotopy class (with fixed endpoints) contains a unique geodesic for any nonpositively curved metric on ∂M (in particular for m, tand $G_{\lambda}(m)$). Given a c-curve γ , we will denote by $L_m(\gamma)$ (respectively $L_t(\gamma)$, $L_{G_{\lambda}}(m)(\gamma)$) the length of that geodesic for the corresponding metric.

Corollary 5.11 Let γ be a *c*-curve in ∂M , then $L_t(\gamma) \leq L_{G_{\lambda}(m)}(\gamma)$.

This follows directly from Lemma 5.9, since any minimizing c-curve in $(S, G_{\lambda}(m))$ has shorter length for t, and the minimizing curve in (S, t) in the same homotopy class is even shorter. Note also that for any c-curve γ , $i(\lambda, \gamma) \leq CL_m(\gamma)$. This follows from Proposition 5.3, and by the fact that the lengths of the c-curves that are segments between two singular points of S are bounded from below.

5.3 An upper bound on the lengths of the curves at infinity

The second step in the proof of Proposition 5.2 is a comparison between the lengths of c-curves in the metrics t and m.

Proposition 5.12 There exists a constant C > 0 (independent of M) such that:

- (1) For each *c*-curve γ in ∂M , $L_t(\gamma) \leq CL_m(\gamma)$.
- (2) For each long tube T in the thin part of $(\partial M, m)$, T might also be a long tube for t, but its length for t is at most its length for m plus C.

The proof uses some simple statements on the geometry of long hyperbolic tubes in (S, t). Recall (see [24]) that the Margulis lemma applies to hyperbolic surfaces with cone singularities of angle at most θ , when $\theta \in (0, \pi)$: there exists a constant c_M , depending on θ only, such that the set of points where the injectivity radius is less than c_M is a disjoint union of cusps, disks centered at a cone singularity, and tubes with core of length less than $2c_M$.

We consider in this subsection a hyperbolic tube T, which can be described as isometric to the set of points at distance at most L (for some L > 0) from the unique simple closed geodesic in the quotient of the hyperbolic plane H^2 by a hyperbolic translation of length l. Moreover l is supposed to be small and L large, so that the lengths of the boundary components of T — which are both equal to $l \cosh(L)$ — are equal to c_M . We call σ the core of T, in other terms the unique simple closed geodesic contained in T, and we denote by σ_M the c_M -neighborhood of σ : the set of points at distance at most c_M from σ in T. We will say that a properly embedded geodesic segment in T goes through T if it enters T by one boundary component and exits by the other. In this case it intersects σ exactly once. This description applies to any properly embedded geodesic segment that enters σ_M . The fact that a segment is properly embedded means that it has its endpoints in ∂T .

Lemma 5.13 There exists a constant C > 0 such that $i(\lambda, \partial \sigma_M) \leq Ce^{-L}$: the intersection of λ with the boundary of σ_M is at most Ce^{-L} .

Proof Any properly embedded geodesic segment in the intersection of the support of λ with T, and going through T intersects σ exactly once, but also each of the two connected components of $\partial \sigma_M$. It follows that the intersection with λ of each of the connected components of $\partial \sigma_M$ is equal to $i(\lambda, \sigma)$. But since the length l of σ satisfies $l \leq c_M e^{-L}$, Proposition 5.3 — applied to long segments that wrap many times around σ — shows that $i(\lambda, \sigma) \leq Ce^{-L}$.

Lemma 5.14 There exists a constant C > 0 such that, if g is a properly embedded geodesic segment in T, then the length of the orthogonal projection on σ of $g \cap (T \setminus \sigma_M)$ is at most C.

Proof If $g \subset \sigma$, then $g \cap (T \setminus \sigma_M) = \emptyset$ and the result applies. We suppose from here on that g is not contained in σ . If g is contained in one connected component of the complement of σ in T then, since g is embedded, its orthogonal projection on σ is injective, so that the length of its orthogonal projection is bounded by the length of σ . Otherwise, it follows from standard hyperbolic geometry arguments that g intersects σ exactly once.

Consider the universal cover \tilde{T} of T; it is isometric to the set of points at distance at most L from a geodesic $\tilde{\sigma} \subset H^2$, which is the lift of σ . Choose one of the connected components, say \tilde{g} , of the lift of g to \tilde{T} . It intersects the lift of $\partial \sigma_M$ with an angle that is bounded from below; otherwise g could not intersect σ . It follows from this, and from elementary geometric properties of the hyperbolic plane, that the length of the orthogonal projection on $\tilde{\sigma}$ of each of the segments of \tilde{g} outside the lift of σ_M is bounded from above by a constant.

Corollary 5.15 There exists a constant C > 0 such that, whenever g_0 is a properly embedded geodesic segment in T such that the orthogonal projection of g_0 on σ is injective, then $i(\lambda_{|T \setminus \sigma_M}, g_0) \leq C$.

Proof Let *c* be a properly embedded geodesic segment in the intersection with *T* of the support of λ , and let *c'* be one of the connected components of $c \cap T \setminus \sigma_M$. Since both *c* and g_0 are geodesic segments, the union of the orthogonal projections on σ of the segments of g_0 and of *c* between two successive intersections between them covers σ .

It follows that the number of intersections between c' and g_0 is at most equal to $(l_{c'} + l_{g_0})/l$, where $l_{c'}$ is the length of the orthogonal projection of c' on σ and l_{g_0} is the length of the orthogonal projection of g_0 on σ (and l is the length of σ).

But the hypothesis on g_0 shows that $l_{g_0} \leq l$, while Lemma 5.14 shows that $l_{c'} \leq C$. So the number of intersections between c' and g_0 is at most Ce^L , where C is some positive constant.

Since this inequality applies to all geodesic segments in the support of λ , we find that

$$i(\lambda_T \setminus \sigma_M, g_0) \le C e^L i(\lambda, \partial \sigma_M)$$

and Lemma 5.13 then shows that $i(\lambda_T \setminus \sigma_M, g_0)$ is bounded by a positive constant. \Box

Proof of Proposition 5.12 Let γ be a c-curve in ∂M . It follows from Proposition 5.3 that

$$L_{G_{\lambda}(m)}(\gamma) \leq L_m(\gamma) + i(\lambda, \gamma) \leq C(L_m(\gamma) + 1),$$

where here again C is a constant depending only on the topology of M. Moreover Corollary 5.11 indicates that

$$L_t(\gamma) \le L_{G_{\lambda}(m)}(\gamma)$$

and point (1) follows.

For point (2) consider a closed geodesic γ contained in the union of T and of the thick part of ∂M , such that:

- The intersection of γ with T has two connected components γ_1 and γ_2 .
- The intersection of γ with the thick part of ∂M (for *m*) has two connected components γ'_1 and γ'_2 , and each has length bounded by *C*.

If T separates the boundary component of ∂M containing it, γ has to go through T twice, otherwise it is not necessary but it is still possible to choose γ with this property, and both cases can then be treated in a uniform manner.

Once such a curve γ has been found, it is possible to change it by Dehn twists so that, in addition to the conditions above, the segments γ_1 and γ_2 "wrap" at most

2354

once around T, ie their orthogonal projection to the core σ of T is injective. This is achieved by "untwisting" γ as much as is necessary.

Denote as above by σ the core of T, and by σ_M the set of points at distance at most c_M from σ . Since γ wraps at most once around T, the length of the intersection of γ_1 and γ_2 with σ_M is at most $3c_M$. It then follows from Proposition 5.3 that

$$i(\lambda_{|\sigma_M}, \gamma) \leq C,$$

where *C* is some positive constant. By the same proposition, the intersection of γ with the restriction of λ to the thick part of $(\partial M, m)$ is at most *C*. But Corollary 5.15 shows that

$$i(\lambda_{|T\setminus\sigma_M},\gamma) \leq C.$$

Putting together those estimates we obtain that $i(\lambda, \gamma) \leq C$, where C is yet another positive constant. The definition of the grafted metric then proves that

$$L_{G_{\lambda}(m)}(\gamma) \leq L_m(\gamma) + C.$$

Finally Lemma 5.9 indicates that the length of γ for *t* is less than that for $G_{\lambda}(m)$. The result follows.

5.4 A bound on the length of the earthquake lamination

We now switch from 3-dimensional to 2-dimensional geometry to show that an upper bound on the length of curves in $(\partial M, t)$ — relative to the length of the same curves in $(\partial M, m)$, as stated in Proposition 5.12 — implies a lower bound on the same lengths. Proposition 5.2 will follow.

We consider a closed surface Σ , with some marked points x_1, \ldots, x_{n_0} , and an angle $\theta_i \in (0, \pi)$ attached to x_i .

Proposition 5.16 For each C > 0 there is a constant C' > 0 as follows. Consider two hyperbolic metrics $h, h' \in \mathcal{H}_{\Sigma, x, \theta}$ such that:

- (1) For each *c*-curve γ in Σ , $L_{h'}(\gamma) \leq CL_h(\gamma)$.
- (2) For each long tube T in the thin part of (Σ, h), T might also be a long tube for h', but its length for h' is at most its length for h plus C.

Let $v \in \mathcal{M}_{\Sigma,x}$ be the measured lamination such that $h' = E_r(v)(h)$. Then the length $L_h(v)$ is at most C'.

The proof of Proposition 5.16 will use a basic estimate on the variation of the length of curves under an earthquake, essentially taken from [13].

Proposition 5.17 Let $m \in \mathcal{ML}_{\Sigma,x}$ be a measured lamination, let $g \in \mathcal{H}_{\Sigma,x,\theta}$ be a hyperbolic metric with cone singularities, and let $g' := E_m^r(g)$. Let γ be a *c*-curve. Then

$$|L_g(\gamma) - L_{g'}(\gamma)| \le i(m, \gamma) \le L_g(\gamma) + L_{g'}(\gamma).$$

Proof The upper bound on $i(m, \gamma)$ can be found in [13, Lemma 7.1, page 76]; it is stated there for closed curves, but the proof extends directly to segments between two singular points.

For the lower bound on $i(m, \gamma)$, suppose first that the support of m is a disjoint union of simple closed curves. Consider the geodesic (for g) γ_0 homotopic to γ in Σ_x , and let γ_1 be its image by the earthquake E'_m , along with the union of the segments in the support of m between two points corresponding — after the earthquake — to one intersection of m with γ_0 . γ_1 is homotopic to γ_0 in Σ_x . Clearly $L_{g'}(\gamma_1) =$ $L_g(\gamma) + i(m, \gamma)$, while $L_{g'}(\gamma) \leq L_{g'}(\gamma_1)$. It follows that $L_{g'}(\gamma) \leq L_g(\gamma) + i(m, \gamma)$. The same inequality also holds with g and g' exchanged, and the lower bound on $i(m, \gamma)$ follows. The result when m is a general lamination — not rational — holds by density of the rational laminations in $\mathcal{ML}_{\Sigma,x}$.

We now return to the notation used in Proposition 5.16. Note that the support of ν is a geodesic lamination in (Σ, g) . It is therefore possible to consider the intersection of ν with the thin (respectively thick) part of Σ for g, which we call ν_t (respectively ν_T). The same decomposition can be done for g', leading to ν'_t and ν'_T .

We first state a basic property of hyperbolic surfaces, which is necessary below.

Lemma 5.18 There exist r > 0, C > 0 and $\theta_0 \in (0, \pi)$, depending only on the supremum θ_M of the θ_i and on the genus of Σ , such that, for any $x \in \Sigma_T$ and any geodesic segment γ_0 of length 2r centered at x, there exists a closed geodesic in Σ of length at most C intersecting γ_0 with angle at least θ_0 .

Proof Note that any maximal segment in the thick part of a topologically nontrivial hyperbolic surface (with cone singularities of angle less than π) intersects some closed geodesic, of length bounded by a constant *C*.

The statement therefore follows from a straightforward compactness argument. Indeed, if the constant $\theta < \theta_M$ did not exist, there would be a sequence of thick hyperbolic surfaces with boundary $\Sigma_{T,n}$ (with cone singularities of angles less than π), for which the optimal value of r would go to infinity, or the optimal value of θ would go to π , as $n \to \infty$. This sequence could be taken of fixed topology, and the diameter of those surfaces would then be bounded, so that r would necessarily be bounded.

We could then choose a converging subsequence, and obtain a thick hyperbolic surface (with cone singularities of angle less than θ_M) for which some maximal geodesic segment intersects no closed geodesic of length less than C transversally, a contradiction.

Lemma 5.19 There exists a constant *C* (depending only on the genus of Σ) so that the length of v_T is at most *C*.

Proof Let $r_0 > 0$ be smaller than the injectivity radius of (Σ, h) at each point of Σ_T . There exists another number $r_1 \in (0, r_0)$ with the following property: if γ_0 and γ_1 are two disjoints geodesics in H^2 and $x \in \gamma_0$ is at distance at most r_1 from γ_1 , then any geodesic intersecting γ_0 at distance less than r_1 from x and making an angle bigger than θ_0 with γ_0 intersects γ_1 at distance at most r_0 from x.

Choose a large constant $C_1 > 0$. If the length of ν_T were bigger than some large constant, the sum of the weights of the segments of the support of ν intersecting some geodesic disk of radius r_1 and center $x \in \text{supp}(\nu) \cap \Sigma_T$ would be bigger than C_1 . Applying the previous lemma, with γ_0 equal to a segment containing x in the support of ν , would yield a closed curve c in Σ_T , of bounded length, such that $i(c, \nu)$ is arbitrarily large.

Proposition 5.17 would then show that the length of c for h' is much larger than the length of c for h, contradicting point (1) in the hypothesis of Proposition 5.16.

Lemma 5.20 There exists a constant C > 0 as follows. Let $\gamma \subset \Sigma_T$ be a geodesic segment of length at most c_M . Then

$$i(\nu, \gamma) \leq CL(\nu_T).$$

Proof We will consider the case when ν is rational; the general case follows by density of the rational measured laminations.

Let *r* be the injectivity radius of Σ_T . Let ν_{γ} be the union of the intersections with Σ_T of all geodesic segments centered at a point $x \in \gamma$, of length 2r, in the support of ν . Each of those segments has length at least *r*, since at least one side of *x* is contained in Σ_T .

By definition of r, those segments intersect γ exactly once. Moreover, the length of ν_T is larger than the sum over the segments of their length times their weight (this sum is finite since ν is rational). But this sum is at least $ri(\nu, \gamma)$, so that $L(\nu_T) \ge ri(\nu, \gamma)$. This proves the lemma.

Lemma 5.21 There exists a constant C > 0 such that, if T is a tube of length 2L in (Σ_t, h) , with core σ , then $i(\nu, \sigma) \leq Ce^{-L}$.

Proof $L_h(\sigma) = c_1 e^{-L}$, where c_1 is some constant. Point (1) in the hypothesis of Proposition 5.16 shows that the length of σ for h' is at most $c_2 e^{-L}$, where c_2 is another positive constant. But Proposition 5.17 then yields the result.

Recall that we call σ_M the set of points at distance at most c_M from σ .

Lemma 5.22 There exists a constant C > 0 such that, if T is a tube of length 2L in (Σ_t, h) , with core σ , then the length for h of the restriction of ν to σ_M is at most C.

Note that there is no reason to believe that this statement is optimal; indeed, it appears quite reasonable to think that the bound could be improved to Ce^{-L} . The bound given here, however, is both sufficient for our needs and easier to obtain.

Proof We know by Lemma 5.21 that $i(\nu, \sigma) \leq C_1 e^{-L}$, where $C_1 > 0$ is some constant. It follows that, if $L_h(\nu_{|\sigma_M})$ is larger than some constant C_2 , then each leaf of ν intersects σ with a very small angle, and for any geodesic segment γ going through T and intersecting σ with angle bigger than $\pi/4$, $i(\gamma, \nu_{|\sigma_M}) \geq C_2 e^L$.

Let γ_1 be a closed geodesic in (Σ, h) that has two segments in Σ_T and two segments going through T. Furthermore we choose γ_1 with the smallest length. Since Σ_T has bounded diameter, there is C > 0 such that $\ell_h(\gamma_1) \leq C + 2T$. Since γ_1 has minimal length, it intersects σ with angle at most $\pi/4$ at each of the two intersections. Then $i(\nu, \gamma_1) \geq 2C_2e^L$ so that, by Proposition 5.17, $L_{h'}(\gamma_1)$ is much larger than $L_h(\gamma_1)$. This contradicts the hypothesis of Proposition 5.16.

Proof of Proposition 5.16 According to Lemma 5.19 the length of the restriction of ν to the thick part of Σ is bounded by a constant (depending only on the genus of Σ). Σ is the union of Σ_T , a finite set of neighborhoods of cusps and cone singularities (which can be disregarded because geodesic laminations do not enter them), and a finite set of long thin tubes, the number of those tubes being at most 3g - 3, where g is the genus of Σ . Let T be one of those tubes and let σ be its core. Then $L(\nu_{|\sigma_M})$ is bounded by a constant by Lemma 5.22. Moreover, the length of each maximal segment of the support of ν in $T \setminus \sigma_M$ is at most $2e^L$, and each is contained in a maximal segment in T (contained in the support of ν) which intersects σ once. Since $i(\nu, \sigma) \leq Ce^L$, the length of the restriction of ν to $T \setminus \sigma_M$ is at most 4C. Summing all contributions to the length of ν yields the desired result.

Proof of Proposition 5.2 The statement clearly follows from Proposition 5.12 and from Proposition 5.16. \Box

6 The conformal structure at infinity

This section contains the proof of Theorems 1.7 and 1.14, mostly as a consequence of Proposition 5.2.

6.1 A topological lemma

We first state a simple topological lemma, necessary below to apply Proposition 5.2 as directly as possible. We fix a closed surface *S* of genus at least 2, a n_0 -tuple of points $x = (x_1, \ldots, x_{n_0})$ and a n_0 -tuple of angles $\theta = (\theta_1, \ldots, \theta_{n_0}) \in (0, \pi)^{n_0}$.

Lemma 6.1 Let c > 0, and let $K \subset \mathcal{H}_{S,x,\theta}$ be a compact subset. The set of all elements of $\mathcal{H}_{S,x,\theta}$ obtained by a right earthquake along a measured lamination of length at most c on an element of K is relatively compact.

Proof Let $m \in K$. The set of measured laminations $l \in \mathcal{ML}$ of length less than c for m is compact in $\mathcal{ML}_{S,x}$. Since the earthquake map is continuous relative to the measured lamination factor, the set

$$E_r(\{l \in \mathcal{ML}_{S,x} \mid L_m(l) \le c\} \times \{m\})$$

is compact in $\mathcal{H}_{S,x,\theta}$.

Again because the earthquake map E_r is continuous, it follows that there is a neighborhood U_m of m in $\mathcal{H}_{S,x,\theta}$ such that the image by E_r of

$$\{(l,m') \in \mathcal{ML}_{S,x} \times U_m \mid L_{m'}(l) \le C\}$$

is relatively compact.

Since K is compact, it is covered by finitely many such neighborhoods U_{m_i} , for m_i in K. The result follows.

6.2 Compactness relative to the conformal structure at infinity

The previous considerations lead to a simple proof of Proposition 1.6 from Proposition 5.2 and Lemma 3.13.

Consider a sequence $(g_n)_{n \in \mathbb{N}}$ of quasifuchsian metrics with particles, as in Proposition 1.6. Let $(m_n)_{n \in \mathbb{N}}$ be the sequence of induced metrics on the boundary of the convex core, and let t_n be the sequence of hyperbolic metrics in the conformal class at infinity τ_n . Since $(\tau_n)_{n \in \mathbb{N}}$ converges by the hypothesis of Proposition 1.6, $(t_n)_{n \in \mathbb{N}}$ converges to a limit t, so it remains in a compact subset of $\mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$.

But m_n is obtained from t_n by an earthquake along a measured lamination v_n which, by Proposition 5.2, has bounded length. Lemma 6.1 therefore shows that $(m_n)_{n \in \mathbb{N}}$ remains in a compact subset of $\mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$. We can therefore extract a subsequence so that $(m_n)_{n \in \mathbb{N}}$ converges.

Lemma 3.13 then shows that $(g_n)_{n \in \mathbb{N}}$ has a subsequence that converges to a quasifuchsian metric with particles. This proves the proposition.

6.3 Proof of Theorem 1.7

We are now ready to prove Theorem 1.7. It is helpful to introduce additional notation:

- $\mathcal{M}_{S,x} := \bigcup_{\theta \in (0,\pi)^N} \mathcal{M}_{S,x,\theta}$ is the space of quasifuchsian metrics with particles on $S \times \mathbb{R}$ with a fixed number of particles but with varying angles.
- $\mathcal{H}_{S,x} := \bigcup_{\theta \in (0,\pi)^N} \mathcal{H}_{S,x,\theta}$ is the space of hyperbolic metrics on S with a fixed number of cone singularities but with varying angles.
- $\Delta_{S,x} := \bigcup_{\theta \in (0,\pi)^N} \mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$ is a kind of diagonal with respect to the angle variable in $\mathcal{H}_{S,x} \times \mathcal{H}_{S,x}$.

Note that, by a result of Troyanov [56] already mentioned above, $\mathcal{H}_{S,x}$ can be naturally identified with $\mathcal{T}_{S,x} \times (0,\pi)^N$. The notation is nonetheless useful in the argument presented here.

Consider the natural map

$$\Phi_{S,x}: \mathcal{M}_{S,x} \to \mathcal{H}_{S,x} \times \mathcal{H}_{S,x}$$

sending a hyperbolic metric with particles on $S \times \mathbb{R}$ (with cone angles given by the θ_i) to the conformal structures at $\pm \infty$. It follows from the definition that the image of $\Phi_{S,x}$ is contained in $\Delta_{S,x}$.

Let $\Phi_{S,x,\theta}$ be the restriction of $\Phi_{S,x}$ to $\mathcal{M}_{S,x,\theta}$, for a fixed $\theta \in (0,\pi)^N$. The main result of [43] is that — in a slightly more general context, allowing for more topology and for singularities along a graph — $\Phi_{S,x,\theta}$ is a local homeomorphism from $\mathcal{M}_{S,x,\theta}$ to $\mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$. It follows that $\Phi_{S,x}$ is a local homeomorphism from $\mathcal{M}_{S,x}$ to $\Delta_{S,x}$. Moreover, $\Phi_{S,x}$ is proper by Proposition 1.6, so that it is a covering of $\mathcal{T}_{S,x,\theta} \times \mathcal{T}_{S,x,\theta}.$

To prove that $\Phi_{S,x}$ is a (global) homeomorphism we need to show that some elements of the target space have exactly one inverse image. Suppose that for all $i \in \{1, ..., N\}$, $\theta_i = 2\pi/k_i$, where $k_i \in \mathbb{N}$, $k_i \ge 2$. Let $\tau_+, \tau_- \in \mathcal{H}_{S,x,\theta}$. There exists a finite covering $\pi: \overline{S} \to S$, with ramification of order k_i at the x_i , such that τ_+ (respectively τ_-) lifts to a nonsingular hyperbolic metric $\overline{\tau}_+$ (respectively $\overline{\tau}_-$). By the Bers double uniformization theorem [8] $\overline{\tau}_+$ and $\overline{\tau}_-$ are in the conformal class at infinity of a unique quasifuchsian hyperbolic metric, say \overline{g} , on $\overline{S} \times \mathbb{R}$. Since it is unique, \overline{g} is invariant under the deck transformations of π , so that \overline{g} is the pull-back to $\overline{S} \times \mathbb{R}$ of a hyperbolic metric g on $S \times \mathbb{R}$, with cone singularities of angle θ_i along $\{x_i\} \times \mathbb{R}, 1 \le i \le N$. This construction also shows that g is unique since any other hyperbolic metric with particles of the given angles would lift to a nonsingular quasifuchsian metric on $\overline{S} \times \mathbb{R}$, which would have to be \overline{g} . This shows that (τ_+, τ_-) has a unique inverse image by $\Phi_{S,x}$, so that $\Phi_{S,x}$ is a homeomorphism from $\mathcal{M}_{S,x}$ to $\Delta_{S,x}$.

6.4 Proof of Theorem 1.14

We need another natural map.

Definition 6.2 Let $\Psi_{S,x,\theta}$: $\mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta} \to \mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$ be defined as follows. Given $(t_+, t_-) \in \mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$ and $\theta = (\theta_1, \ldots, \theta_{n_0}) \in (0, \pi)^n$, there is by Theorem 1.7 a unique quasifuchsian metric with particles $g \in \mathcal{M}_{S,x,\theta}$ such that we have $\Phi_{S,x,\theta}(g) = ([t_+], [t_-])$. Then

$$\Psi_{S,x,\theta}(t_+,t_-) = (m_+,m_-),$$

where m_+ and m_- are the conformal classes of the induced metrics on the upper and lower boundary components of the convex core of $(S \times \mathbb{R}, g)$, respectively.

According to Proposition 5.2 and Lemma 6.1, if $(m_+, m_-) = \Psi_{S,x,\theta}(t_+, t_-)$, then m_+ is C_q -quasi-conformal to t_+ , and m_- is C_q -quasi-conformal to t_- . This shows that $\Psi_{S,x,\theta}$ is proper and extends continuously to a map that is the identity on the boundary at infinity of $\mathcal{H}_{S,x,\theta} \times \mathcal{H}_{S,x,\theta}$, so that it is onto. This proves Theorem 1.14.

7 Some questions and remarks

7.1 Some questions

The quasifuchsian cone-manifolds described above are direct extensions of the "usual" quasifuchsian hyperbolic manifolds, which have received much attention over the last couple of decades. It is quite natural to wonder whether some properties that have been conjectured in the smooth case can be extended to the singular setting.

Question 7.1 Does uniqueness hold in Theorem 1.14?

Geometry & Topology, Volume 18 (2014)

Another natural question, which is "dual" to the previous one in a precise sense (see Schlenker [47]) concerns the measured bending lamination on the boundary of the convex core.

Question 7.2 Does uniqueness hold in Theorem 1.13?

The same questions can be asked for submanifolds of quasifuchsian cone-manifolds that are convex but have a smooth boundary, which is orthogonal to the singular locus. In the smooth case it is known (Schlenker [48]) that one can prescribe the induced metric on the boundary, as well as the its third fundamental form (the smooth analog of the measured bending lamination) and that each is obtained uniquely, it would be interesting to know whether the same is true for quasi-fuchsian cone-manifold. The methods of [48] do not appear to extend directly to the singular case.

Since the Ahlfors–Bers theorem extends as Theorem 1.7 to quasifuchsian manifolds with particles, it is quite natural to ask whether the ending lamination conjecture (see Brock, Canary and Minsky [17] and Brock, Masur and Minsky [18]) can also be extended to hyperbolic manifolds with particles. A natural starting point would be to consider manifolds homeomorphic to $S \times \mathbb{R}$, where S is a closed surface of genus at least 2.

Note also that those questions are not necessarily restricted to quasifuchsian conemanifolds, and could also be asked for "convex cocompact cone-manifolds", if that term is understood in a proper way.

7.2 AdS manifolds with particles

Mess [3; 40] discovered a remarkable analogy between quasifuchsian hyperbolic 3manifolds and globally hyperbolic maximal compact (GHMC) AdS manifolds. In particular he proved an analog of the Bers double uniformization theorem form GHMC AdS manifolds: on a manifold homeomorphic to $S \times \mathbb{R}$, where S is a closed surface of genus at least 2, the space of GHMC AdS manifolds is parametrized by the product of two copies of the Teichmüller space of S, through the "left" and "right" parts of the holonomy representation.

GHMC AdS manifolds also have a convex core, whose boundary has a hyperbolic induced metric, as in the quasifuchsian setting, and is pleated along a measured lamination. The analog of Theorem 1.13 holds in that context [14]: any two measured laminations that fill S can be obtained as the bending lamination of the boundary of the convex core. But the uniqueness remains elusive, as in the quasifuchsian setting.

Moreover, the analog of Theorem 1.14 is also conjectured to be true but no result is known.

It is also possible to consider GHMC AdS manifolds with "particles", ie cone singularities along time-like geodesics, for which the angle is less than π . The analog of the Bers double uniformization theorem (more directly, the analog of Theorem 1.7) holds in this AdS setting [13]. The analog of Theorem 1.13 is also true in that setting [14]. However no analog of Theorem 1.14 is known.

Still in the AdS setting, new phenomena can occur when the singularity is along a graph (so that the particles are allowed to interact); see [5; 6]. One can associate to a GHMC AdS manifold with a graph of interacting particles a sequence of pairs of points in the Teichmüller space of the underlying surface, with each pair corresponding to a "slice" where no interaction occurs. It would be interesting to know whether any analog of this description holds for quasifuchsian hyperbolic manifolds with cone singularities along a graph (perhaps with some conditions on the cone angles, for instance cone angles less than 2π).

7.3 The renormalized volume

Theorem 1.7 for quasifuchsian manifolds with particles has possible applications to the Teichmüller theory of hyperbolic surfaces with cone singularities (of fixed angles) on a surface. Indeed it was remarked in [33] that the definition of the renormalized volume of a quasifuchsian 3-manifolds extends to manifolds with particles. Knowing Theorem 1.7, it is possible to remark that the key property of the renormalized volume — to be a Kähler potential for the Weil-Petersson metric on Teichmüller space — extends to the natural Weil-Petersson metric on the Teichmüller space of hyperbolic metrics with cone singularities (of prescribed angle less than π) on a surface; the proof from [33; 35] directly extends to this setting.

One direct consequence is that this Weil–Petersson metric is Kähler, as was discovered by Schumacher and Trapani [49] by other means. This metric, however, seems to depend on the choice of the cone angles.

Another possible application is to some properties of the grafting map considered on hyperbolic surfaces with cone singularities of angle less than π , as considered in [34; 35]. This is however less directly related to what we are doing here, since it only uses the geometry of 3-dimensional hyperbolic ends—rather than quasifuchsian metrics—with particles.

Appendix A: Quasiconformal estimates

This appendix contains the proof of Proposition 1.15. The first step is a simple extension to hyperbolic surfaces with cone singularities of some classical tools concerning pants decompositions.

A.1 Pants decompositions

The content of this subsection is probably well known; see [24] for closely related considerations. We include this material for completeness.

Let S be a closed surface, and let h be a hyperbolic metric on S with cone singularities at some points x_1, \ldots, x_{n_0} , with cone angles $\theta_1, \ldots, \theta_{n_0} \in (0, \pi)$. If h had cusps or geodesic boundary components — at the x_i rather than cone singularities, it would be quite natural to consider pants decompositions of (S, h). With cone singularities of angles less than π , it remains possible.

Definition A.1 A *singular pair of pants* is a hyperbolic surface with geodesic boundary, possibly containing cone singularities of angle less than π , which is either:

- A hyperbolic pair of pants (with geodesic boundary) containing no cone singularity.
- A hyperbolic annulus with geodesic boundary containing exactly one cone singularity.
- A hyperbolic disk with geodesic boundary containing exactly two cone singularities.

Given a singular hyperbolic pair of pants, its three geodesic boundary components or cone singularities will be called its *legs*. We hope that the reader will excuse us for this weird and perhaps confusing terminology.

Definition A.2 A *pants decomposition* of *S* is a decomposition $S = S_1 \cup \cdots \cup S_n$ of *S* as the union of finitely many subsurfaces with disjoint interior, each of which is a singular pair of pants.

It is implicit in this definition that the boundary of the S_i contains no cone singularities; the cone singularities are each contained in the interior of one of the singular pairs of pants.

Lemma A.3 There exists a constant $C_p > 0$ such that, for any choice of *S* and *h*, (S, h) has a pants decomposition with all boundary curves of length less than C_p .

Sketch of the proof A standard recursive argument (see Benedetti and Petronio [7]) reduces the proof to showing that, for any hyperbolic surface with cone singularities (of angle less than π) and geodesic boundary, there is a simple closed geodesic of length at most C_p that is not homotopic to a singular point or to a boundary component.

This in turn follows from other standard arguments, for instance based on comparing the area of the surface (given by a suitable Gauss–Bonnet formula; see [56]) to the area of embedded geodesic disks. \Box

Definition A.4 Let P be a singular pant. Its *leg invariants* are the length of its geodesic boundary components and the angles at its cone singularities.

For instance, the boundary invariants of a (nonsingular) hyperbolic pair of pants are the lengths of its boundary components.

Lemma A.5 Each hyperbolic pair of pants is uniquely determined, up to isometry, by its leg invariants and by the type of its "legs", whether they are boundary components or cone singularities.

The proof follows the classical arguments used for nonsingular hyperbolic pairs of pants; it is based on elementary properties of some hyperideal hyperbolic triangles stated below in three propositions (the first two have probably been known since Lobachevsky).

Recall that a hyperideal triangle can be defined, using the projective model of the hyperbolic plane, as a triangle that might have its vertices either in the hyperbolic plane, on its ideal boundary, or outside the closure of the hyperbolic plane (considered as the interior of a disk in the projective plane), but with all edges intersecting the hyperbolic plane. A vertex is then ideal if it is on the ideal boundary, and strictly hyperideal if it is outside the closed disk.

Recall also that given a point v_0 outside the closure of the projective model of H^2 (in the projective plane), there is a unique hyperbolic geodesic, v_0^* , such that any the intersection with the projective model of H^2 of any projective line containing v_0 is orthogonal to v_0^* . This geodesic is called the line *dual* to v_0 .

We introduce here a slightly restricted notion of hyperideal triangle.

Definition A.6 An *extended hyperbolic triangle* is a hyperbolic triangle with one or more strictly hyperideal vertices and its other vertices in the "interior" of the hyperbolic plane. A *truncated hyperbolic triangle* is the intersection of an extended hyperbolic triangle with the hyperbolic half-planes bounded by the lines dual to its strictly hyperideal vertices (and not containing the endpoints of the edges going towards those vertices).

For instance, a right-angle hyperbolic hexagon can be considered — in two ways — as a truncated hyperbolic triangle, with three strictly hyperideal vertices. Given a hyperbolic triangle, its *angles* are the hyperbolic angles at the nonhyperideal vertices and the length of its intersections with the lines dual to the strictly hyperideal vertices. Note that those lengths can quite naturally be considered as angles (they are then imaginary numbers) but it is not necessary to enter such considerations here (see Schlenker [45; 46] for more details).

There is a natural way to define the *edge lengths* of an extended hyperbolic triangle. The length of the edge joining two vertices v and v' is:

- The hyperbolic distance between v and v', if neither v nor v' is strictly hyperideal.
- The hyperbolic distance between v and the line dual to v', when v' is hyperideal but v is not.
- The distance between the lines dual to v and v', when both are strictly hyperideal.

It is useful to remark that the lengths and angles of an extended hyperbolic triangle satisfy a natural extension of the cosine formula. Moreover it is quite easy to check that an extended hyperbolic triangle, with vertices of given type, is uniquely determined by two lengths and one angle, or by two angles and one length.

The three next lemmas are easy extensions of classical results and their proofs are left to the reader.

Lemma A.7 An extended hyperbolic triangle is uniquely determined by the type of its vertices — whether they are "usual" or strictly hyperideal vertices — and its edge lengths.

Lemma A.8 An extended hyperbolic triangle is uniquely determined by the type of its vertices — whether they are "usual" or strictly hyperideal vertices — and its angles. The angles at the "usual" vertices can take any value in $(0, \pi/2)$, while the "angles" at the strictly hyperideal vertices can be any numbers in $(0, \infty)$.

Lemma A.9 Each singular pair of pants has a unique decomposition as the union of two copies of a truncated hyperbolic triangle (glued along their common boundary).

Proof of Lemma A.5 By Lemma A.8, the two extended triangles glued to obtain a hyperbolic pair of pants are uniquely determined by their angles, which can take any value as long as the angles at the "usual" vertices are less than $\pi/2$. This shows that hyperbolic pairs of pants are uniquely determined by their leg invariants, and any values are possible as long as the angles at the singular points are less than π .

2366

We now turn to the parametrization of hyperbolic metrics with cone singularities by Fenchel–Nielsen type coordinates. We first state a lemma on the existence and uniqueness of a pants decomposition from topological data, leaving the proof to the reader since it is the same as in the nonsingular case.

Lemma A.10 A pants decomposition is uniquely determined by the choice of the boundary curves $\gamma_1, \ldots, \gamma_N$, considered as simple closed curves in $S \setminus \{x_1, \ldots, x_{n_0}\}$, under the hypothesis that:

- The γ_i can be realized as pairwise disjoint curves.
- Each connected component of their complement is either a pair of pants containing none of the x_i , or a cylinder containing exactly one of the x_i , or a disk containing exactly two of the x_i .

Finally we state the main consequence, on the parametrization of hyperbolic metrics with cone singularities of fixed angle by Fenchel–Nielsen coordinates, again leaving the proof to the reader. Note that the Dehn twist parameters are defined only in a relative way, however this is exactly the same as in the nonsingular case (see [7]).

Corollary A.11 Given a (topological) pants decomposition of *S* with boundary curves $\gamma_1, \ldots, \gamma_N$, there is a homeomorphism

$$\mathcal{T}_{S,x,\theta} \to (\mathbb{R}_{>0} \times \mathbb{R})^N$$

sending a hyperbolic metric to the length and fractional Dehn twist parameters at the γ_i .

The fractional Dehn twist parameters used here are the translation length of one side with respect to the other so that, for a boundary curve of length l, a parameter equal to l corresponds to a "usual" Dehn twist (the other possibility is to use an "angle" parameter, where 2π corresponds to full Dehn twist).

A.2 Proof of Proposition 1.15

It is now possible to use the pants decomposition provided by Lemma A.3 to prove Proposition 1.15: the induced metric on the boundary of the convex core is (uniformly) quasi-conformal to the conformal structure at infinity.

The starting point is that a pants decomposition of $(\partial M, m)$ with boundary curves of bounded length defines a pants decomposition of $(\partial M, \tau)$ with boundary curves of approximately the same length. Recall that the constant C_p was introduced in Lemma A.3. **Lemma A.12** There exists a constant C > 0 as follows. Let $\gamma_1, \ldots, \gamma_N$ be simple closed curves on ∂M , defining a pants decomposition, of lengths less than C_p for m. Then

for all
$$i \in \{1, \ldots, N\}$$
, $\frac{L_m(\gamma_i)}{C} \leq L_\tau(\gamma_i) \leq CL_m(\gamma_i)$.

Proof The upper bound is a direct consequence of the first point in Proposition 5.12. If γ_i is short for m — ie it is the core of a long tube in the thin part of $(\partial M, m)$ — then the second point of Proposition 5.12 proves the lower bound for γ_i .

Suppose now that γ_i is realized in $(\partial M, m)$ as a closed geodesic in the thick part of ∂M . Then there exists a closed geodesic γ' intersecting γ_i of length at most C_p . If the length of γ_i in $(\partial M, \tau)$ were small, than γ_i would be realized in $(\partial M, \tau)$ as the core of a long tube T in the thin part of $(\partial M, \tau)$. But then γ' would have to be long (at least as long as the T). This would contradict the first point in Proposition 5.12, and this proves the lower bound for γ_i .

Lemma A.13 There exists a constant C > 0 such that, for each of the γ_i , the difference in the Dehn twist parameter corresponding to γ_i in *m* and in τ is at most $C(|\log(L_m(\gamma_i))| + 1)$.

The precise form of the estimate is important only if γ_i is short for *m* (and therefore for τ), in which case $|\log(L_m(\gamma_i))|$ is half the length of the tube in the thin part of $(\partial M, m)$ containing γ_i .

Proof Suppose first that γ_i is not short. Then it is contained in the thick part of $(\partial M, m)$, and there exists another curve γ' , intersecting γ_i , of uniformly bounded length. A Dehn twist parameter bigger than some constant would extend the length of γ' by more than is allowed by Proposition 5.12; this proves the lemma in this first case.

The same argument can be used when γ_i is short (ie when it is the core of a long thin tube): then, γ' can be chosen to have length bounded by a constant time $|\log(L_m(\gamma_i))|$, and this defines the maximal Dehn twist parameter along γ_i .

Proof of Proposition 1.15 Let $\gamma_1, \ldots, \gamma_N$ be disjoint closed curves, defining a pants decomposition of $(\partial M, m)$ with boundary curves of length less than C_p , as in Lemma A.3. Let l_1, \ldots, l_N be the length of the γ_i for m, and let d_1, \ldots, d_N be the Dehn twist parameters for the same curves.

Lemma A.10 shows that the γ_i also define a pant decomposition of $(\partial M, \tau)$, let l'_i be the length of the γ_i for τ , and let d'_i be their Dehn twist parameters. Lemma A.12

indicates that the l'_i are within a fixed multiplicative constant of the l_i , while by Lemma A.13

(3)
$$|d'_i - d_i| \le C(|\log(l_i)| + 1),$$

where C is some positive constant.

Let m' be the hyperbolic metric with cone singularities obtained by gluing pairs of pants with boundary lengths equal to the l_i , but with Dehn twist parameters equal to the d'_i .

Note that m' is C_1 -quasi-conformal to m, for some uniform constant $C_1 > 0$. To prove this, remark that for each $i \in \{1, ..., N\}$ the set of points at distance at most $C(|\log(L_m(\gamma_i))| + c_M)$ from γ_i is an annulus, and that those annuli are disjoint. One can therefore build a C_1 -quasi-conformal diffeomorphism between m and m' that is an isometry in the complement of those annuli around the γ_i , and which is "twisted" in those annuli, with a twisting parameter that is an affine function of the distance to the γ_i .

The second and last step is to show that m' is C_3 -quasi-conformal τ . Since those two metrics differ only by the lengths of the boundary curves γ_i , and in view of Equation (3), this follows again from a simple and explicit construction, which we leave to the interested reader.

Note that it might be possible to prove Proposition 1.15 using the same type of arguments as those used by Epstein and Marden [26] in the nonsingular case. This would have the advantage of providing directly a quasiconformal constant independent on the genus of the boundary.

Acknowledgements

We are grateful to an anonymous referee for many remarks which were most useful in improving the exposition of the arguments presented here.

This work was partially supported by the ANR program ETTT (ANR-09-BLAN-0116-01). CL is partially supported by the ANR programs Groupes and ETTT (ANR-09-BLAN-0116-01). JS is partially supported by the ANR program ETTT (ANR-09-BLAN-0116-01). The authors acknowledge support from U.S. National Science Foundation grants DMS 1107452, 1107263, 1107367 "RNMS: GEometric structures And Representation varieties" (the GEAR Network).

References

- L Ahlfors, L Bers, *Riemann's mapping theorem for variable metrics*, Ann. of Math. 72 (1960) 385–404 MR0115006
- [2] A D Aleksandrov, V A Zalgaller, *Intrinsic geometry of surfaces*, Trans. Math. Monog. 15, Amer. Math. Soc. (1967) MR0216434
- [3] L Andersson, T Barbot, R Benedetti, F Bonsante, W M Goldman, F Labourie, K P Scannell, J-M Schlenker, Notes on: "Lorentz spacetimes of constant curvature" by G Mess, Geom. Dedicata 126 (2007) 47–70 MR2328922
- [4] W Ballmann, M Gromov, V Schroeder, Manifolds of nonpositive curvature, Progress in Mathematics 61, Birkhäuser, Boston (1985) MR823981
- [5] T Barbot, F Bonsante, J-M Schlenker, Collisions of particles in locally AdS spacetimes, I: Local description and global examples, Comm. Math. Phys. 308 (2011) 147–200 MR2842974
- [6] T Barbot, F Bonsante, J-M Schlenker, Collisions of particles in locally AdS spacetimes, II: Moduli of globally hyperbolic spaces, Comm. Math. Phys. 327 (2014) 691–735 MR3192047
- [7] R Benedetti, C Petronio, Lectures on hyperbolic geometry, Springer, Berlin (1992) MR1219310
- [8] L Bers, Simultaneous uniformization, Bull. Amer. Math. Soc. 66 (1960) 94–97 MR0111834
- [9] AL Besse, Einstein manifolds, Ergeb. Math. Grenzgeb. 10, Springer, Berlin (1987) MR867684
- [10] M Boileau, B Leeb, J Porti, Geometrization of 3-dimensional orbifolds, Ann. of Math. 162 (2005) 195–290 MR2178962
- [11] M Boileau, J Porti, Geometrization of 3-orbifolds of cyclic type, Astérisque 272, Soc. Math. France, Paris (2001) 208 MR1844891
- [12] F Bonahon, J-P Otal, Laminations measurées de plissage des variétés hyperboliques de dimension 3, Ann. of Math. 160 (2004) 1013–1055 MR2144972
- [13] F Bonsante, J-M Schlenker, AdS manifolds with particles and earthquakes on singular surfaces, Geom. Funct. Anal. 19 (2009) 41–82 MR2507219
- F Bonsante, J-M Schlenker, Fixed points of compositions of earthquakes, Duke Math. J. 161 (2012) 1011–1054 MR2913100
- [15] M Bridgeman, Average bending of convex pleated planes in hyperbolic three-space, Invent. Math. 132 (1998) 381–391 MR1621436
- [16] M Bridgeman, R D Canary, From the boundary of the convex core to the conformal boundary, Geom. Dedicata 96 (2003) 211–240 MR1956842

Geometry & Topology, Volume 18 (2014)

- [17] JF Brock, RD Canary, YN Minsky, The classification of Kleinian surface groups, II: The ending lamination conjecture, Ann. of Math. 176 (2012) 1–149 MR2925381
- [18] JF Brock, H Masur, Y N Minsky, Asymptotics of Weil–Petersson geodesic, I: Ending laminations, recurrence, and flows, Geom. Funct. Anal. 19 (2010) 1229–1257 MR2585573
- [19] K Bromberg, Hyperbolic cone-manifolds, short geodesics, and Schwarzian derivatives, J. Amer. Math. Soc. 17 (2004) 783–826 MR2083468
- [20] K Bromberg, Rigidity of geometrically finite hyperbolic cone-manifolds, Geom. Dedicata 105 (2004) 143–170 MR2057249
- [21] R Brooks, J P Matelski, Collars in Kleinian groups, Duke Math. J. 49 (1982) 163–182 MR650375
- [22] A J Casson, S A Bleiler, Automorphisms of surfaces after Nielsen and Thurston, London Math. Soc. Student Texts 9, Cambridge Univ. Press (1988) MR964685
- [23] D Cooper, CD Hodgson, SP Kerckhoff, Three-dimensional orbifolds and conemanifolds, MSJ Memoirs 5, Math. Soc. Japan, Tokyo (2000) MR1778789
- [24] E B Dryden, H Parlier, Collars and partitions of hyperbolic cone-surfaces, Geom. Dedicata 127 (2007) 139–149 MR2338522
- [25] D Dumas, Complex projective structures, from: "Handbook of Teichmüller theory, Vol. II", (A Papadopoulos, editor), IRMA Lect. Math. Theor. Phys. 13, Eur. Math. Soc., Zürich (2009) 455–508 MR2497780
- [26] DBA Epstein, A Marden, Convex hulls in hyperbolic space, a theorem of Sullivan, and measured pleated surfaces, from: "Analytical and geometric aspects of hyperbolic space", (DBA Epstein, editor), London Math. Soc. Lecture Note Ser. 111, Cambridge Univ. Press (1987) 113–253 MR903852
- [27] **D B A Epstein**, **V Markovic**, *The logarithmic spiral: A counterexample to the* K = 2 *conjecture*, Ann. of Math. 161 (2005) 925–957 MR2153403
- [28] JL Harer, RC Penner, Combinatorics of train tracks, Annals of Mathematics Studies 125, Princeton Univ. Press (1992) MR1144770
- [29] C D Hodgson, S P Kerckhoff, Rigidity of hyperbolic cone-manifolds and hyperbolic Dehn surgery, J. Differential Geom. 48 (1998) 1–59 MR1622600
- [30] G 't Hooft, The evolution of gravitating point particles in 2 + 1 dimensions, Classical Quantum Gravity 10 (1993) 1023–1038 MR1219446
- [31] G 't Hooft, Quantization of point particles in (2 + 1)-dimensional gravity and spacetime discreteness, Classical Quantum Gravity 13 (1996) 1023–1039 MR1390098
- [32] S P Kerckhoff, The Nielsen realization problem, Ann. of Math. 117 (1983) 235–265 MR690845

- [33] K Krasnov, J-M Schlenker, On the renormalized volume of hyperbolic 3–manifolds, Comm. Math. Phys. 279 (2008) 637–668 MR2386723
- [34] K Krasnov, J-M Schlenker, A symplectic map between hyperbolic and complex Teichmüller theory, Duke Math. J. 150 (2009) 331–356 MR2569616
- [35] K Krasnov, J-M Schlenker, The Weil–Petersson metric and the renormalized volume of hyperbolic 3–manifolds, from: "Handbook of Teichmüller theory, Vol. III", (A Papadopoulos, editor), IRMA Lect. Math. Theor. Phys. 17, Eur. Math. Soc., Zürich (2012) 779–819 MR2952776
- [36] R S Kulkarni, U Pinkall, A canonical metric for Möbius structures and its applications, Math. Z. 216 (1994) 89–129 MR1273468
- [37] F Labourie, Métriques prescrites sur le bord des variétés hyperboliques de dimension
 3, J. Differential Geom. 35 (1992) 609–626 MR1163450
- [38] C Lecuire, Plissage des variétés hyperboliques de dimension 3, Invent. Math. 164 (2006) 85–141 MR2207784
- [39] C T McMullen, The moduli space of Riemann surfaces is Kähler hyperbolic, Ann. of Math. 151 (2000) 327–357 MR1745010
- [40] G Mess, Lorentz spacetimes of constant curvature, Geom. Dedicata 126 (2007) 3–45 MR2328921
- [41] R Meyerhoff, A lower bound for the volume of hyperbolic 3-manifolds, Canad. J. Math. 39 (1987) 1038–1056 MR918586
- [42] YN Minsky, The classification of punctured-torus groups, Ann. of Math. 149 (1999) 559–626 MR1689341
- [43] S Moroianu, J-M Schlenker, Quasi-Fuchsian manifolds with particles, J. Differential Geom. 83 (2009) 75–129 MR2545031
- [44] J-P Otal, Le théorème d'hyperbolisation pour les variétés fibrées de dimension 3, Astérisque 235, Soc. Math. France, Paris (1996) MR1402300
- [45] J-M Schlenker, Métriques sur les polyèdres hyperboliques convexes, J. Differential Geom. 48 (1998) 323–405 MR1630178
- [46] J-M Schlenker, Convex polyhedra in Lorentzian space-forms, Asian J. Math. 5 (2001) 327–363 MR1868937
- [47] J-M Schlenker, Des immersions isométriques de surfaces aux variétés hyperboliques à bord convexe, Sémin. Théor. Spectr. Géom. 21, Univ. Grenoble I, Saint-Martin-d'Hères (2003) 165–216 MR2052831
- [48] J-M Schlenker, Hyperbolic manifolds with convex boundary, Invent. Math. 163 (2006) 109–169 MR2208419
- [49] G Schumacher, S Trapani, Weil–Petersson geometry for families of hyperbolic conical Riemann surfaces, Michigan Math. J. 60 (2011) 3–33 MR2785861

- [50] **M Spivak**, *A comprehensive introduction to differential geometry, Vol. III*, 2nd edition, Publish or Perish, Wilmington, DE (1979)
- [51] L A Takhtajan, L-P Teo, Liouville action and Weil–Petersson metric on deformation spaces, global Kleinian reciprocity and holography, Comm. Math. Phys. 239 (2003) 183–240 MR1997440
- [52] L A Takhtajan, P G Zograf, On the uniformization of Riemann surfaces and on the Weil–Petersson metric on the Teichmüller and Schottky spaces, Mat. Sb. 132(174) (1987) 304–321, 444 MR889594 In Russian; translated in Math. USSR-Sb. 60 (1988) 297–313
- [53] L A Takhtajan, P G Zograf, Hyperbolic 2–spheres with conical singularities, accessory parameters and Kähler metrics on $\mathcal{M}_{0,n}$, Trans. Amer. Math. Soc. 355 (2003) 1857–1867 MR1953529
- [54] **W P Thurston**, *The geometry and topology of 3-manifolds*, lecture notes, Princeton University (1978–1981) Available at http://library.msri.org/books/gt3m
- [55] WP Thurston, Earthquakes in two-dimensional hyperbolic geometry, from: "Lowdimensional topology and Kleinian groups", (D B A Epstein, editor), London Math. Soc. Lecture Note Ser. 112, Cambridge Univ. Press (1986) 91–112 MR903860
- [56] M Troyanov, Prescribing curvature on compact surfaces with conical singularities, Trans. Amer. Math. Soc. 324 (1991) 793–821 MR1005085
- [57] F Waldhausen, Eine Verallgemeinerung des Schleifensatzes, Topology 6 (1967) 501– 504 MR0220300

Institut de Mathématiques de Toulouse, UMR CNRS 5219, Université Toulouse III 31062 Toulouse Cedex 9, France

Mathematics Research Unit, BLG, University of Luxembourg, Campus Kirchberg 6 rue Richard Coudenhove-Kalergi, L-1359 Luxembourg, Luxembourg

lecuire@math.univ-toulouse.fr, jean-marc.schlenker@uni.lu

Proposed: Benson Farb Seconded: David Gabai, Jean-Pierre Otal Received: 18 February 2013 Accepted: 5 October 2013